

A Service of

ZBU

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Gibbons, Eric M.; Mukhopadhyay, Sankar

# Article New evidence on international transferability of human capital

IZA Journal of Development and Migration

# Provided in Cooperation with:

IZA – Institute of Labor Economics

*Suggested Citation:* Gibbons, Eric M.; Mukhopadhyay, Sankar (2020) : New evidence on international transferability of human capital, IZA Journal of Development and Migration, ISSN 2520-1786, Sciendo, Warsaw, Vol. 11, Iss. 1, pp. 1-39, https://doi.org/10.2478/izajodm-2020-0009

This Version is available at: https://hdl.handle.net/10419/298680

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



Eric M. Gibbons\* and Sankar Mukhopadhyay\*

# New Evidence on International Transferability of Human Capital

# Abstract

This article sheds new light on the portability of human capital. We estimate the returns to source country experiences, viz., general, occupation-specific, and task-specific experiences, using data from the New Immigrant Survey (NIS), conducted in 2003. While the "returns to general experience" has been discussed in the literature, we are not aware of any previous attempt to estimate the returns to source country occupation-specific and task-specific experiences. Our estimates show that even though the returns to source country general experience is negligible, returns to source country occupation-specific experience is economically and statistically significant. We also find that returns to source country abstract (specifically analytical) task-specific experience is substantial and significant. Our results are robust to inclusion of source country wage, which may reflect unobservable characteristics that influence wages. We explore whether returns to work experience vary by income level in the source country or by an immigrant's skill level.

Current version:	April 01, 2020
Keywords:	occupation-specific experience, task-specific experience,
	general experience, source country, immigration
JEL codes:	J3, J61, J62
Corresponding authors:	Eric M. Gibbons
	gibbons.183@osu.edu
	Sankar Mukhopadhyay
	sankarm@unr.edu

1 Morrill Hall, The Ohio State University at Marion, 1465 Mt. Vernon Avenue, Marion, OH 43302

2 Department of Economics, University of Nevada, Reno, 1664 N. Virginia Street, Reno, NV 89557

<sup>©</sup> The Author(s). 2020. Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. Cite as: Gibbons and Mukhopadhyay. *IZA Journal of Development and Migration* (2020) 11:9.

# 1 Introduction

The number of immigrants worldwide has increased at a dramatic rate in the past few decades. Grieco and Trevelyan (2010) have reported that the number of foreign-born individuals in the U.S. has increased from 9.6 million (or about 4.7% of the U.S. population) in 1970 to 38 million (or about 12.5% of the U.S. population) in 2009. The economic outcomes of immigrants may depend on the international portability of human capital acquired in their source countries. Current literature suggests that the return to source country education is generally positive (Friedberg, 2000; Schaafsma and Sweetman, 2001; Bratsberg and Ragan, 2002; Ferrer and Riddell, 2008; Chiswick and Miller, 2008; Nielson, 2011; Ferrer et al., 2006). On the other hand, the return to source country work experience is small or even negligible (Friedberg, 2000; Cohen-Goldner and Eckstein, 2008; Basilio et al., 2017; Green and Worswick, 2002; Aydemir and Skuterud, 2005).

The literature on international portability has almost exclusively focused on general human capital (education and general work experience [GE]). However, a number of papers in the past three decades have stressed that a large part of human capital is either industry specific or occupation specific, or both (Neal, 1995; Parent, 2000; Dustmann and Meghir, 2005; Kambourov and Manovskii, 2009; Williams, 2009; Sullivan, 2010). More recently, Gibbons and Waldman (2004), Gathmann and Schönberg (2010), and Schulz et al. (2013) have pointed out the importance of task-specific human capital. These papers have established the importance of both occupation-specific experience (OE) and task-specific experience (TE) in wage growth. However, how specific experiences (OE and TE) acquired by immigrants in their source countries are valued in a host country remain unexplored.

In this paper, we use data from the New Immigrant Survey (Guillermina et al. 2006) to estimate the association between source country GE, OE, and TE of immigrants and their U.S. wages in their first U.S. job. We are not aware of any other studies that have estimated how OE and TE acquired in the source countries of the immigrants are valued in a host country.

We estimate wage regressions where GE, OE, and TE simultaneously enter the wage equation in a quartic manner (Murphy and Welch, 1990), along with an extensive set of control variables, including source country and U.S. entry year fixed effects. We check the sensitivity of returns to GE, OE, and TE to different levels of occupational granularity – from the one-digit (least granular) to the three-digit level (most granular) – and different levels of task granularity – from three types of tasks (Abstract, Routine, and Manual) to five types of tasks [nonroutine cognitive: analytical (NRCA), nonroutine cognitive: interpersonal (NRCI), routine cognitive (RCOG), routine manual (RMAN), and nonroutine manual: physical adaptability (NRMN)] (Autor et al., 2003; Autor and Handel, 2013).

Ordinary least squares (OLS) estimates show that an immigrant with 5 years of OE earns between 20.3% and 27.2% more than an immigrant without any OE. The return to OE is the highest when we define occupations in the most granular way. We also find that the return to Abstract TE is positive and significant. An immigrant with 5 years of Abstract TE earns between 16.2% and 20.2% more than a comparable immigrant without any Abstract TE. Returns to Routine and Manual TE are always insignificant. A more disaggregated task-based approach shows that among the abstract tasks, analytical (NRCA) skills are transferable. Other abstract skills, including interpersonal (NRCI) skills, have limited or no transferability. The remainder of the paper has the following structure. Section 2 discusses the relevant literature, Section 3 presents the data, Section 4 presents the empirical results, and Section 5 concludes.

# 2 Background

This paper relates to two strands of literature. The first strand is about international portability of human capital. In an influential paper, Friedberg (2000) shows that returns to source country GE are much smaller than returns to host country GE, or even insignificant for immigrants in Israel. Cohen-Goldner and Eckstein (2008) report similar results. Basilio et al. (2017) find small but statistically significant returns to source country GE. Kossoudji (1989) finds negligible returns to source country GE in the case of immigrants to the U.S. Green and Worswick (2002) report similar results for Canada. Aydemir and Skuterud (2005) and Green (2009) report that even though immigrants to Canada enjoyed a substantial return to source country GE in the early 1980s, the return to source country GE declined to zero by the early 1990s. Aydemir and Skuterud (2005) further find that most of the decline was due to a larger presence in Canada of immigrants from "nontraditional" (mostly developing countries) countries. Worswick (2003) suggests that even though the return to source country work experience is negligible at the time of entry, source country work experience may become valuable after immigrants spend time in the host country. However, Chiswick et al. (2005) and Beenstock et al. (2009) find evidence to the contrary.

The second strand relates to the literature on the estimation of OE/TE. Previous studies on returns to specific human capital have established that there are significant wage returns to OE/TE. Many of these studies have noted that OE/TE may be endogenous in the wage equation. Unobserved occupational and firm match, as well as unobserved ability, may bias returns to both OE and TE (Gathmann and Schönberg, 2010). A good-quality occupational or firm match will likely cause returns to OE to be upward biased, whereas, a poor match in one or both job dimensions may cause the worker to switch occupations, resulting in a downward bias on the return to OE and TE. Workers with higher unobserved ability are expected to have greater occupational or firm attachment and will likely yield upward-biased OLS estimates on OE and TE.

Altonji and Shakotko (1987) propose an instrumental variable (IV) approach to account for selection into occupations, which has been used by Kambourov and Manovskii (2009), Sullivan (2010), Williams (2009). Gathmann and Schönberg (2010) apply local labor market and occupational-similarity instruments to address selection on TE. Workers switching occupations are more likely to choose an occupation similar in task composition and have a higher propensity to remain in the same occupation if there are greater employment opportunities in that market. Unfortunately, IV is not feasible with our analysis due to data constraints.

Previous studies on return to experience have encountered similar difficulties using immigrant data. Reinhold and Thom (2013) discuss the endogeneity problems in estimating returns to U.S. experience for workers returning to Mexico. The authors point to similar constraints in constructing instruments using the Mexican Migration Project (MMP). The MMP, like the NIS, contains cross-sectional data and prohibits the construction of the instrument proposed by Altonji and Shakotko (1987). The authors rely on OLS to estimate the return to U.S.

experience using various control strategies. Similar to Reinhold and Thom (2013), we proceed with OLS, utilizing an extensive set of controls to estimate returns to source country experiences. As a robustness check, we control for time-constant heterogeneity by adding, as a control, the wages earned by the immigrants in their source countries before moving to the U.S.

Existing literature notes that OLS and IV estimates on the return to OE are not only qualitatively similar, but the direction of bias is unclear. For example, Sullivan (2010) reports that 5 years of OE (at the one-digit occupational level) increases wages by 10.36% using OLS and 13.30% using IV estimates. Kambourov and Manovskii (2009) report that the return to 5 years of OE (at the two-digit occupation level) is 10.98% using OLS and 8.75% using IV estimates. Dustmann and Meghir (2005) report that one additional year of sector tenure increases wages by 1.4% using OLS and 1.0% using IV estimates. Gathmann and Schönberg (2010) find that the OLS estimates on the return to TE are smaller than the IV estimates. Their findings suggest that OLS estimates may provide a lower bound to returns on TE.

There are some additional complications that are somewhat specific to studies on immigrant earnings. First, we are using data on immigrants and not workers who lost their jobs due to plant closures. Therefore, they may not be what Neal (1995) called "exogenously displaced". An individual will immigrate to the U.S. if the expected (purchasing power parity [PPP] adjusted) wage is sufficiently higher (compared to the cost of migrating) in the U.S. compared to that in his/her home country. However, the differences in wage levels between the U.S. and most of the source countries do not necessarily imply that an immigrant achieves a better job match (Chiswick and Miller, 2008). The second issue is how to disentangle the effects of U.S. experience and entry cohort heterogeneity. Fortunately, the NIS includes information on the wage in their first job in the U.S. We use this wage because, at that point, immigrants do not have any U.S. work experience. Another added advantage of using immigrants' first U.S. wage is that job tenure is zero for all immigrants (Topel, 1991). In addition, we use entry year dummies to allow for entry cohort heterogeneity.

The third issue is that immigrants may face occupational downgrading after moving to the host country (Bauer and Zimmerman, 1999; Akresh, 2008; Chiswick and Miller, 2008). In this context, there is no clear consensus on the relationship between immigrants' skill and occupational downgrading. For example, Bauer and Zimmerman (1999) show that higher-skilled immigrants face sharper occupational downgrading immediately after immigration to Germany. On the other hand, Akresh (2008) shows that more-skilled immigrants experience very little occupational downgrading after migrating to the U.S. Chiswick and Miller (2008) find that greater preimmigration labor market experience leads to poorer job matches in the host country. In a recent paper, Imai et al. (2017) find substantial occupational downgrading among immigrants to Canada. They show that this occupational downgrading leads to reduced wages. They also find that this downgrading, and the consequential reduction in wages, is greater among immigrants with limited language ability. However, these papers do not address the issue of transferability of OE and TE, which is the primary focus of this paper.

# 3 Data and descriptive statistics

We use data from the NIS, 2003. The NIS provides extensive information on new immigrants to the U.S., who received their permanent residency (green card) in 2002–2003. Thus, as part of the survey, 8573 adult immigrants completed interviews. If the spouse of a principal immigrant was also an immigrant, then the spouse was also interviewed. In all, 4915 spouses were interviewed. Therefore, the total number of foreign-born immigrants interviewed who received their green card between 2002 and 2003 was 13488. Our study design requires us to focus on immigrants who worked in their source countries before immigrating to the U.S. and worked in the U.S. after immigrating. Out of 13488 immigrants, 6977 (59%) held at least one job in their source countries before moving to the U.S. We focus on only those immigrants who were between the ages of 16 and 65 years (both included) at the time of immigrating to the U.S. This reduces the sample size to 5919 immigrants. We further restrict the sample based on the availability of covariates, those who report participating in the U.S. labor market after immigrating, and those who have an accurate work history in their source countries. The sample size after accounting for the availability of covariates and labor market participation is 2713. However, approximately 9% of immigrants in this sample have work histories where either their stated start date in their first source country job occurs after the conclusion of their last source country job or there is an overlap between start and end dates between first and last source country jobs. We omit these observations from our analysis, which leaves a sample of 2460 for immigrants who participated in the U.S. labor market.

Before we discuss the descriptive statistics, we briefly describe the representativeness aspect of the sample. Since the NIS only interviewed lawful permanent residents (LPRs), it may not be representative of all immigrants to the U.S. However, a comparison with the 2003 immigrant sample using the current population survey (CPS) suggests that average age and gender

Characteristics	CP	S		Ν	lis
	Full sample <sup>a</sup>	<b>Employed</b> <sup>b</sup>	Full sample <sup>c</sup>	Our sample	Our sample
					w/o employment principal
Age	37.32	35.89	37.38	37.41	37.51
	(14.25)	(11.07)	(10.77)	(9.12)	(9.48)
Male	0.52	0.64	0.49	0.60	0.58
	(0.50)	(0.48)	(0.50)	(0.49)	(0.59)
Education (in years)	11.14	11.59	13.22	14.38	13.88
	(4.52)	(4.46)	(4.35)	(4.27)	(4.17)
% Employed <sup>†</sup>	0.63	1.0	0.62	1.0	1.0
	(0.48)		(0.49)		
Hourly wage		<b>2.23</b> <sup>††</sup>		2.33	2.18
		(0.87)		(0.67)	(0.59)
Observations	13078	7521	11601	2460	1976

Table 1 Sample comparisons between 2003 CPS noncitizen respondents and NIS respondents

Note: Standard deviations are reported in parentheses.

<sup>a</sup>Noncitizens' age 16+ years. <sup>b</sup>Noncitizens who report being employed and having a strictly positive annual income. <sup>c</sup>NIS full sample includes immigrants between the ages of 16–65 years at the time of entry into the U.S. after accounting for available covariates.

<sup>†</sup>Civilian employment.

<sup>††</sup>Hourly wage is calculated from a respondent's previous year's annual salary, winsorized at 1%. The reported value is the natural log of the derived hourly wage.

compositions are similar in the CPS and the NIS (see Table 1). However, respondents in the NIS sample are more educated and earn about 10% more in wages. When we exclude immigrants with employer-sponsored LPRs (employment principal), there is little difference in the average wage between the two samples. Therefore, as a robustness check, we exclude the employment principals. Our results are robust to such changes (see Section 4.3).

In the NIS, respondents answered questions regarding their work history in their source countries. They were asked about the starting dates, ending dates, and occupations in their first (after they turned 16) and last (if worked in more than one job) jobs in their source countries. For each job, occupations are reported using 2002 Census three-digit categories. Following standard practices in the literature, we classify occupations in one-, two-, and three-digit Standard Occupational Classification (SOC) coding to construct OE for immigrants.

To construct TE, we first generate task measures following Acemoglu and Autor (2011) using occupational characteristics from O\*NET (2003), herein ONET. We include different aggregations of tasks to explore the importance of granularity of task definitions. We use the five task definitions of Acemoglu and Autor (2011) and the more common three-task definitions of Autor et al. (2003).<sup>1</sup>

We take U.S.-based task measures and generate ranks of tasks in an occupation. We do this based on the findings of Dicarlo et al. (2016), who show that skill ranks across countries are similar despite differences in production technologies. The relationship between skills and tasks are that workers apply their stock of skills to perform tasks that produce some measure of output (Autor, 2013). Therefore, if the skill profiles of occupations in the source countries are similar to those in the U.S., then the types of tasks performed in a source country may be similar to those in the U.S. The use of U.S. task data for foreign occupations may pose challenges due to differences in production technologies across countries. However, if task ranks (based on the ones that are more pronounced in an occupation) are similar across countries, then we can use U.S. occupational task ranks as a hierarchical rule for occupations in foreign countries. To construct the TE, we take the most predominant task in a source country occupation and allow experience in that task to transfer, if and only if, the next occupational match is predominant in the same task.<sup>2</sup>

Therefore, TE is only transferable in the predominant occupational task. As an example illustrating the intuition of this methodology, take a worker employed as a welder in his/her source country. This occupation is likely to consist mainly of the welder performing a routine task (welding), but the secondary task may be more manual in nature. The predominant task

<sup>1</sup> The five tasks are used to construct the Abstract, Routine, and Manual tasks; see the Appendix of the paper by Autor and Handel (2013) for more details on the construction process.

<sup>2</sup> Since we are using U.S. task measures to proxy for source country tasks, we perform robustness checks to our measure of TE using two different approaches in how we define the transferability of task-specific human capital. First, we follow the methodology proposed by Gathmann and Schönberg (2010), constructing a one-dimensional distance measure (angular separation distance metric) for each occupational pair. Under this approach, TE is equal to  $(1-\arg \operatorname{sep}_{ol} ) T_o$ + $(1-\arg \operatorname{sep}_{ol} _{o} US) T_o I$ , where  $T_o$  denotes tenure in the first occupation o and  $T_o I$  denotes tenure in the last occupation. Otherwise, TE is equal to OE for those workers who never switch occupations. The second approach uses the rank of ONET tasks to construct  $k \times oo$  measures of distance. To illustrate how TE is constructed using task rank, let us suppose that a worker's first occupation in source country is dominated by an Abstract task and the last occupations in the Abstract task dimension is distance  $_{koo}I = |1 - 2|/K = 1$ , where K denotes the number of tasks. The transferability metric is defined as  $(1 - \operatorname{distance}_{koo}I / K)$ . The worker in this example, where K = 3, will be able to transfer two-thirds of his/her Abstract-specific task experience from the first job to the second. TE for the worker at the time of employment in his/ her first U.S. job is  $TE_{ko}I = (1 - \operatorname{distance}_{koo}I/K)T_o + (1 - \operatorname{distance}_{ko}I_o US/K)T_o I$ . Estimates on returns to specific experience using our alternative measures of TE are qualitatively similar to those presented in the main results.

in the same occupation in the U.S. is likely to be similar. However, a welder in the U.S. may perform an abstract task (using software to design welds) as a secondary task. Therefore, the task ranks comprising secondary, tertiary, and so on in a source country may not be the same as in the U.S.

As discussed above, the selection into occupations may result in upward or downward bias of the OLS estimates on the returns to OE, depending on a worker's unobservables. Return to OE/TE is likely to be biased upward for immigrants who stay in the same occupation upon entry into the U.S. labor market. In contrast, returns to OE/TE for those who switch occupations upon entry may be biased downward. In the NIS, we are able to identify a worker as a stayer, switcher, or nonparticipant in the job market. We estimate each labor market participant type separately to show the possible upper and lower bounds of the returns to OE and TE. Table 2 displays the types of immigrant workers in our NIS sample. We categorize workers as stayers if they were employed in the same occupation in their source country as their first occupation in the U.S. (Type I) or if their last occupation in the granularity of occupational classification, the numbers of immigrants classified as stayers are 1006, 694, and 551 at the one-, two-, and three-digit occupational coding levels (total number of workers in our sample is 2460). A worker is categorized as a switcher if his/her last occupation in the source country differs from his/her first U.S. occupation.

There are three important takeaways from Table 2. First, we can separately estimate returns to specific experience for stayers and switchers. Second, identification of OE from GE is based on workers being employed in multiple source country occupations prior to immigrating (Type I vs Type II). Third, OE in the sample of workers who switched occupations is identified by analyzing workers employed in the same U.S. occupation as their first occupation in the source country (Type I). Furthermore, when we estimate returns to specific experience for the entire NIS sample, the existence of Type I switchers ensures that OE is not solely identified by stayers.

In the NIS, questions regarding immigrants' source country work experience were limited to their first and last jobs. Any gaps in their work history between their first and last jobs could consist of employment in first occupation, employment in last occupation, employment in another occupation that is neither their first nor their last occupation, unemployment, or nonparticipation in the labor market. We assume that any gap in their work history consists of employment when deriving GE. For consistency purposes, if a respondent had the same occupation in their first and last jobs in the source country, their OE includes the gap

Occupation in first source country job	Occupation in last source country job	Occupation in first U.S. job	Classification	Туре
Х	Х	Х	Stayer	I
Х	Y	Y	Stayer	Ш
Х	Y	Х	Switcher	I
Х	Х	Y	Switcher	Ш
Х	Y	Z	Switcher	Ш
			Not working	Nonparticipant

 Table 2
 NIS interview questions and classification as stayer, switcher, and nonparticipant

Figure 1 Average experience by sample and digit. Note: (a) General and occupationspecific experience. (b) Task-specific experience.



(number of years) as part of their work experience.<sup>3</sup> Type II stayers' OE is the number of years employed in their last job. OE for Type I switchers is defined as the number of years worked in their first and only job.

Figure 1 reports the mean values of GE, OE, and TE by sample (all, stayers, and switchers) and occupational granularity (digit level). Switchers, on average, have more GE than stayers. The average GE at the two-digit occupational coding level is 11.06 years for stayers and 12.30 years for switchers. There are large differences in OE across groups. Stayers, on average, have 8.19 years of OE, whereas switchers have only 0.23 years of OE at the two-digit level. The pattern of occupational stayers having more OE than switchers is the same as discussed in Neal (1995).

Table 3 presents the summary statistics for all immigrants (Column 1), stayers (Columns 2–4), and switchers (Columns 5–7). There are three samples of stayers (and switchers) because

<sup>3</sup> The average gap in work history for stayers is 5.01, 4.85, and 4.61 years at the one-, two-, and three-digit levels respectively.

Parameters	All workers		Stayers			Switchers	;
		1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
Hourly wage	13.589	18.087	19.674	21.34	10.477	11.197	11.35
	(15.882)	(19.697)	(19.717)	(20.23)	(11.610)	(13.359)	(13.59)
Education	13.820	14.375	14.725	14.898	13.437	13.465	13.509
	(4.000)	(3.969)	(3.751)	(3.817)	(3.975)	(4.038)	(3.997)
English (very good)	0.411	0.533	0.597	0.613	0.327	0.338	0.352
	(0.492)	(0.499)	(0.491)	(0.487)	(0.469)	(0.473)	(0.478)
English (good)	0.140	0.132	0.120	0.118	0.145	0.148	0.146
	(0.347)	(0.339)	(0.325)	(0.323)	(0.352)	(0.355)	(0.353)
English (average)	0.057	0.036	0.029	0.031	0.072	0.069	0.065
	(0.232)	(0.186)	(0.167)	(0.173)	(0.259)	(0.253)	(0.247)
English (poor)	0.031	0.029	0.024	0.022	0.033	0.034	0.034
	(0.174)	(0.167)	(0.155)	(0.146)	(0.179)	(0.181)	(0.181)
English (very poor)	0.361	0.270	0.231	0.216	0.423	0.412	0.402
	(0.480)	(0.444)	(0.421)	(0.412)	(0.494)	(0.492)	(0.490)
Female	0.402	0.377	0.395	0.376	0.420	0.405	0.410
	(0.490)	(0.485)	(0.489)	(0.485)	(0.494)	(0.491)	(0.492)
Employment principal	0.197	0.305	0.373	0.437	0.122	0.127	0.127
	(0.398)	(0.461)	(0.484)	(0.497)	(0.327)	(0.333)	(0.333)
Spouse of US citizen	0.113	0.098	0.089	0.078	0.124	0.123	0.124
	(0.317)	(0.298)	(0.285)	(0.268)	(0.329)	(0.328)	(0.329)
Diversity principal	0.122	0.083	0.053	0.051	0.150	0.149	0.143
	(0.328)	(0.275)	(0.225)	(0.220)	(0.357)	(0.357)	(0.350)
Spouse of principal immigrant	0.308	0.304	0.301	0.279	0.310	0.310	0.316
	(0.462)	(0.460)	(0.459)	(0.449)	(0.463)	(0.463)	(0.465)
Region of residence (east)	0.369	0.387	0.379	0.377	0.356	0.365	0.366
	(0.483)	(0.487)	(0.485)	(0.485)	(0.479)	(0.481)	(0.482)
Region of residence (west)	0.275	0.271	0.261	0.260	0.278	0.281	0.280
	(0.447)	(0.445)	(0.439)	(0.439)	(0.448)	(0.450)	(0.449)
Region of residence (south)	0.184	0.183	0.192	0.176	0.185	0.181	0.186
	(0.388)	(0.387)	(0.394)	(0.381)	(0.388)	(0.385)	(0.390)
Years between last source job and US job	1.924	1.744	1.643	1.421	2.049	2.035	2.069
	(3.709)	(3.633)	(3.693)	(3.130)	(3.758)	(3.711)	(3.849)
Workers	2,460	1,006	694	551	1,454	1,766	1,909

## Table 3Summary statistics

Note: Standard deviations are denoted in parentheses.

we define stayers (and switchers) based on the one-, two-, and three-digit occupational coding. Among all the workers, 40.2% are female. The percentage of female workers varies between 37.6% and 39.5% among the stayers and between 40.5% and 42% in the switcher sample. The average number of years of education among all workers is 13.82 years. Stayers have slightly more education than switchers. Average education varies between 14.9 and 14.38 years among

the stayer samples, and it varies between 13.44 and 13.51 years for the switcher samples. Stayers are more proficient in English than are switchers. In our data, between 53.3% and 61.3% of stayers have "very good" English-speaking skills, whereas, among the switchers, between 32.7% and 35.2% have English proficiency categorized as "very good". We use interviewer-reported measure for fluency in spoken English, which is available for all interviewees. The measure takes a value of 1–4, with "1" indicating very good English fluency, and "4" indicating poor English. The English proficiency measure is given a value of "5" if the interview was conducted in a language other than English because the interviewees were not comfortable answering questions in English.

We classify the immigrant source countries into two groups: low income and high income. The countries and regions with a per capita income  $\geq$ 30% compared to the U.S. are classified as high income (Canada, South Korea, Europe and Central Asia, and Poland).<sup>4</sup> The high-income group comprises 23.74% of the entire sample. Therefore, the majority of immigrants are from low-income countries. Stayers, at the two-digit level, consist of approximately the same proportion as the entire sample (23.34%).

Immigrants in this sample have received their green cards in one of the following categories: employment based, spouse of a U.S. citizen, diversity, refugee or asylee, spouse of principal immigrant, and others (family based). There are differences in the distribution of visa types for stayer and switcher samples. For example, between 30.5% and 43.7% of stayer (between 12.2% and 12.7% of switcher) samples are employment principals (their employer sponsored their green card).<sup>5</sup> Finally, in wage regressions, we control for the number of years between the last source country job and the first U.S. job to account for any potential depreciation of human capital.

The primary outcome variable of interest is the (log of) hourly wage of the immigrant's first job in the U.S. – winsorized (replacement of values in the bottom and top tails (outliers) of the distribution equal to the next-nearest value) at the 1% level to reduce the influence of outliers. The winsorized wage distribution has a minimum hourly wage of \$1.93 and maximum hourly wage of \$153.60. All wages are in 2003 prices. The average hourly wage is between \$18.09 and \$21.34 for stayers and between \$10.48 and \$11.35 for switchers across the three levels of occupational aggregation.<sup>6</sup>

## 4 Results

We begin with kernel-weighted local polynomial regressions with (log of) first U.S. wage against source country GE. We focus our discussion on immigrants with  $\leq$ 15 years of source country GE. About 70% of immigrants in our sample have  $\leq$ 15 ears of source country GE. Figure 2 displays the results at each of the three levels of occupational granularity that determines the classification of an immigrant as either a stayer or a switcher. Panel (a) is for one-digit, Panel (b) is for two-digit, and Panel (c) is for three-digit occupational granularity.

<sup>4</sup> We use 2000 per capita GDP to compare income across countries (Source: http://databank.worldbank.org/data/views/ reports/tableview.aspx).

<sup>5</sup> In Figure S1 in Supplementary Materials, we show results from kernel-weighted local polynomial regressions of wages of immigrants' first U.S. job against source country GE after excluding employment principals.

<sup>6</sup> The average hourly wage when employment principals are excluded from the sample is \$11.22. The average hourly wage for stayers and switchers in this subsample is between \$14.77–\$18.27 and \$9.28–\$9.91, respectively.

Figure 2 Kernel-weighted local polynomial – returns to general experience. Notes: (a) One-digit occupational level. (b) Two-digit occupational level. (c) Three-digit occupational level.



Each panel shows three lines: the bold line represents all immigrants, the dash-dot line represents stayers (those who did not switch occupation after immigrating to the U.S.), and the dashed line represents switchers. In each panel, the dashed line is almost flat, suggesting that the switchers receive little to no return to their source country GE. The dash-dot line (stayers), on

the other hand, is relatively steep, suggesting a positive association between wages and source country GE (especially for the first 5 years or so) among immigrants who stay in their source country occupation after immigrating to the U.S. The bold line (all immigrants combined) is a combination of these two groups. The results displayed in each of the panels in Figure 1 suggest a positive association between U.S. wage and source country GE for those immigrant workers classified as a stayer. The variation in returns to GE across occupational classifications, especially early in a worker's career, demonstrate that source country OE may be an important determinant of U.S. wage. We further explore these issues below in subsections 4.1 and 4.2.

#### 4.1 Returns to GE

We start by estimating the following wage regression (nonlinear terms not shown in the equation to reduce clutter).

$$w_i = \alpha + \beta G E_i + X_i \psi + \varepsilon_i \tag{1}$$

The dependent variable,  $w_i$ , is the (log of) hourly wage rate of immigrant *i* in the first U.S. job. The vector of individual characteristics is denoted by  $X_i$  and includes the immigrant's years of education in source country, gender, proficiency in English, visa type, number of years between last source country job and first U.S. job, region of residence in the U.S., source country fixed effects, and U.S. entry year fixed effects. Since previous literature on transferability of source country work experience has only included returns to GE, we start with a specification where we include only GE. We use a flexible functional form with up to the fourth order work experience terms, as suggested by Murphy and Welch (1990). They show that standard (quadratic) Mincer earnings function may understate the early career earnings growth and may overstate midcareer earnings growth.<sup>7</sup> In our regressions, we divide squared, cubed, and quartic terms by 1000 to rescale the coefficients.

Table 4 presents the results based on the first U.S. wage regressions. We estimate this equation for all workers (Column 1) and then separately for stayers (Columns 2–4) and switchers (Columns 5–7). The classification of who is a stayer (or switcher) depends on the granularity of occupational definitions. When we define occupation at the one-digit level, we have 1006 stayers and 1454 switchers. At the two-digit level, we have 694 stayers and 1766 switchers. At the three-digit level, we have 551 stayers and 1909 switchers.

We first discuss the estimates for all workers (Column 1). All the control variables have expected signs. For example, each year of source country education increases wage by 3.4%. Female workers earn 11.2% less than male workers. Workers with "very good" English-speaking skills earn 16.7% more than individuals interviewed in a language different from English (presumably because the interviewees were not comfortable in answering questions in English). Next, we discuss returns to work experience.

Since GE enters the wage equation in a quartic way, we present the cumulative returns to 3, 5, and 10 years of GE in order to discuss the magnitude of the association. Table 5 reports the cumulative return estimates to GE. We present the returns for all workers (Column 1) and then separate the samples into stayers (Columns 2–4) and switchers (Columns 5–7). Estimates for all workers (Column 1) suggest that a worker with 3 years of GE will earn 10.6% more than

<sup>7</sup> For more discussion on the Mincer earnings function, see Mincer (1974).

Parameters	All		Stayers			Switchers	
		1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
GE	0.0441***	0.0671**	0.0968**	0.1091**	0.0220	0.0182	0.0191
	(0.0165)	(0.0308)	(0.0394)	(0.0461)	(0.0183)	(0.0177)	(0.0170)
Square	-3.1734**	-5.3253*	-7.8664**	-8.4873*	-1.1747	-0.8613	-1.0127
	(1.5939)	(3.0278)	(3.9854)	(4.6013)	(1.7558)	(1.7095)	(1.6480)
Cubic	0.0798	0.1577	0.2429	0.2601	0.0149	0.0053	0.0117
	(0.0572)	(0.1100)	(0.1493)	(0.1714)	(0.0627)	(0.0608)	(0.0590)
Quartic	-0.0007	-0.0016	-0.0026	-0.0029	0.0000	0.0001	0.0001
	(0.0007)	(0.0013)	(0.0018)	(0.0021)	(0.0007)	(0.0007)	(0.0007)
Female	-0.1146***	-0.1306***	-0.1394***	-0.1811***	-0.0833***	-0.1000***	-0.0954***
	(0.0234)	(0.0421)	(0.0505)	(0.0589)	(0.0270)	(0.0255)	(0.0246)
Education	0.0343***	0.0425***	0.0395***	0.0421***	0.0282***	0.0313***	0.0312***
	(0.0035)	(0.0066)	(0.0089)	(0.0111)	(0.0040)	(0.0037)	(0.0036)
English: very good	0.1901***	0.2190***	0.3175***	0.3378***	0.1424***	0.1196***	0.1397***
	(0.0370)	(0.0653)	(0.0858)	(0.1110)	(0.0424)	(0.0399)	(0.0382)
English: good	0.0464	0.1140*	0.2325***	0.3001***	0.0098	0.0101	0.0127
	(0.0382)	(0.0631)	(0.0854)	(0.1075)	(0.0478)	(0.0432)	(0.0409)
English: average	0.0030	0.1244	0.1690	0.1922	0.0323	0.0036	0.0074
	(0.0431)	(0.1023)	(0.1462)	(0.1663)	(0.0450)	(0.0445)	(0.0434)
English: poor	-0.0051	-0.0613	0.0757	-0.0095	0.0339	-0.0210	-0.0014
	(0.0468)	(0.0953)	(0.1242)	(0.1597)	(0.0494)	(0.0480)	(0.0450)
Constant	1.8596***	1.5168***	1.5945***	1.4258***	2.2338***	1.9325***	1.9323***
	(0.1590)	(0.2438)	(0.2741)	(0.2974)	(0.2412)	(0.2461)	(0.2362)
Other controls							
Work gap	Х	Х	Х	Х	Х	Х	Х
Visa type	Х	Х	Х	Х	Х	Х	Х
Region	Х	Х	Х	Х	Х	Х	Х
Entry year	Х	Х	Х	Х	Х	Х	Х
Source FE	Х	Х	Х	Х	Х	Х	Х
Ν	2,459	1,006	694	551	1,453	1,765	1,908
<i>R</i> <sup>2</sup>	0.41	0.47	0.48	0.48	0.30	0.32	0.32

Table 4 OLS: returns to source country general experience (quartic)

Note: Robust standard errors are reported in parentheses.

\*0.10, \*\*0.05, and \*\*\*0.01 denote significance levels.

Control "Work gap" denotes the time between immigration and employment in first U.S. job.

Visa types: employment based, spouse of an U.S. citizen, diversity, refugee or asylee, spouse of principal immigrant, and other (family based).

a worker without any GE. The cumulative effects of 5 and 10 years of GE are 15.1% and 19.7%, respectively. All estimates are significant at the 1% level.

Columns 2–4 in Table 5 show the returns to GE for stayers. Estimates suggest that returns to GE are economically and statistically significant among stayers. For example, return to 3 years of GE is between 15.7% and 25.8% for an immigrant who remains in his/her source

Parameters	All		Stayers			Switchers	
		1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.1059***	0.1574**	0.2260***	0.2576**	0.0557	0.0471	0.0486
	(0.0371)	(0.0688)	(0.0874)	(0.1026)	(0.0412)	(0.0399)	(0.0383)
5 Years of GE	0.1508***	0.2208**	0.3161***	0.3638***	0.0823	0.0704	0.0718
	(0.0506)	(0.0933)	(0.1180)	(0.1391)	(0.0564)	(0.0544)	(0.0523)
10 Years of GE	0.1969***	0.2792**	0.3980***	0.4734***	0.1175*	0.1029	0.1021
	(0.0606)	(0.1100)	(0.1378)	(0.1639)	(0.0683)	(0.0651)	(0.0627)
Ν	2,459	1,006	694	551	1,453	1,765	1,908

 Table 5
 OLS estimates of cumulative returns to source country general experience (quartic)

Note: Standard errors are in parentheses.

\*0.10, \*\*0.05, and \*\*\*0.01 denote significance levels.

country occupation after immigrating to the U.S. The corresponding return to 5 years of GE is between 22.1% and 36.4% (between 27.9% and 47.2% for 10 years). The returns are highest when we define stayer in the most granular way (they remain in the same three-digit occupation after immigrating to the U.S.). On the other hand, the return to GE is small and almost never significant for those who switch occupations after immigrating to the U.S. (Columns 5–7). The negligible return to GE among switchers is likely driving the relatively low returns to GE in the sample of all workers (Column 1). The results confirm that the univariate association between source country work experience and U.S. wage presented in Figure 2 survives adding an extensive set of control variables. The evidence presented here suggests that low returns to GE reported in the previous literature (discussed in Section 2) may be a result of the large fraction of immigrants switching occupations after immigrating to a host country. In our data, we find that between 59.1% (at one-digit occupational level) and 77.6% (at three-digit occupational level) immigrants switch occupation after immigrating to the U.S.

#### 4.2 Returns to Experience GE, OE, and TE

Given the evidence that immigrants who stay in their source country occupation receive significant return to their source country work experience, we explore which component(s) of source country work experience are portable and which are not. To that effect, we estimate a regression equation with GE, OE, and TE (nonlinear terms not shown in the equation to reduce clutter) (Gathmann and Schönberg, 2010).

$$w_i = \alpha + \beta G E_i + \delta O E_i + T E_i \gamma + X_i \psi + \varepsilon_i$$
<sup>(2)</sup>

Each type of experience enters the wage equation in a quartic functional form. In our baseline analysis, we use three-task measures (Abstract, Routine, and Manual) to create source country TE (Autor et al., 2003; Autor and Handel, 2013). We also conduct parallel analysis using five-task measures (NRCA, NRCI, RCOG, RMAN, and NRMN) using the definitions in Acemoglu and Autor (2011). We briefly mention, but do not discuss, estimates on the returns to the five-task measures in detail in the text of this paper as the results provide similar conclusions. We report cumulative return estimates using the five-task approach in Tables S2-S7 in Supplementary Materials.

**Figure 3** Joint returns to experience – two-digit level. Returns to 10 years of GE and 0 years of OE and TE are not statistically significant at any of the conventional levels for the samples of "All immigrants", "Stayers", and "Switchers". Returns to 10 years of GE, 3 years of OE, and no TE are significant at the 1% and 10% levels for the "All immigrants" and "Stayer" samples, respectively. Returns to 10 years of GE, 3 years of OE, and no TE are not significant at any of the conventional levels for the Switcher sample. Return to 10 years of GE, 3 years of OE, and 3 years of Abstract TE is significant at the 1%, 5%, and 1% levels for the "All immigrants", "Stayers", and "Switcher" samples, respectively.



Again, we estimate this regression equation for all workers and then separately for stayers and switchers. In each case, we estimate the returns to various types of work experience for occupations categorized at the one-, two-, and three-digit levels. In regressions, we include the same set of control variables reported in Table 4.<sup>8</sup>

To focus our discussion on cumulative returns to GE, OE, and TE, we ask the following question: how much more would an immigrant with 10 years of GE, 3 years of two-digit OE, and 3 years of abstract TE earn compared to an otherwise-comparable immigrant without any source country work experience? We present these results in Figure 3. For this illustration, we use estimates at two-digit occupational categories. We focus on two-digit occupational categories because the previous literature in this area has noted that one-digit categories may be too coarse and three-digit aggregation may be too noisy or produce small subsamples (Mellow and Sider, 1983; Cortes and Gallipoli, 2017). Analysis at the two-digit level may also help with measurement error due to recall problems, which are more prevalent in retrospective surveys (Kambourov and Manovskii, 2012; Moscarini and Thomsson, 2007).

Our estimates suggest that an immigrant with 10 years of GE, but no OE or TE would earn 7.4% (not significant) more than an immigrant without any work experience. An immigrant

<sup>8</sup> Tables SI in Supplementary Materials present the details on the returns to GE, OE, and TE. We do not discuss the details in the text. Comparing estimates from the three different levels of occupational granularity suggests that some of the results are common across all samples and specifications. However, in other cases, not only does the statistical significance vary, the economic significance varies as well. One result that is always true is that once we account for OE and TE, return to GE is never significant. The return to OE is significant at all levels of occupational granularity among all workers, but only at the two-digit level for stayers and three-digit level among switchers. Return to Abstract TE is almost always significant across all samples, but returns to other types of TE (Routine and Manual) are not significant. One potential reason for such inconsistencies regarding returns to OE may be the misclassification of occupations due to the retrospective nature of the survey. Such misclassification may be more important when there is less variation to start with (within stayer and switcher samples). However, this is not unique to this paper.

with 10 years of GE, 3 years of OE, and no TE would earn 24.4% (significant at 1%) more than an immigrant without any work experience. An immigrant with 10 years of GE, 3 years of OE, and 3 years of Abstract TE would earn 34.4% (significant at 1%) more than an immigrant without any work experience. Our results for stayers (Figure 3) suggest that a stayer with 10 years of GE, no OE, and no TE would earn 4.0% (but not significantly) more than an immigrant without any work experience. A stayer with 10 years of GE, 3 years of OE, and no TE would earn 30.2% (significant at 10%) more than an immigrant without any work experience. A stayer with 10 years of GE, 3 years of OE, and 3 years of Abstract TE would earn 36.1% (significant at 5%) more than an immigrant without any work experience. Our results for switchers (Figure 3) suggest that a switcher with 10 years of GE, no OE, and no TE would earn 4.9% (not significant) more than an immigrant without any work experience. A switcher with 10 years of GE, 3 years of OE, and no TE would earn 13.8% (not significant) more than an immigrant without any work experience. A switcher with 10 years of GE, 3 years of OE, and 3 years of Abstract TE would earn 26.7% (significant at 1%) more than an immigrant without any work experience. Therefore, these results suggest that even the switchers may receive economically and statistically significant return if they have OE and Abstract TE.

To represent the results regarding the returns to GE, OE, and TE in an alternative way, we ask the following question: how much more would an immigrant with average work experience earn compared to an immigrant without any work experience. When we consider all immigrants in our sample, an average immigrant has about 12 years of GE, 2 years of OE, 2 years of Abstract TE, 1 year of Routine TE, and 1 year of Manual TE (all rounded to the nearest integer). An immigrant with such a work experience profile will earn 24.5% more (significant at 1%) more than an immigrant without any work experience (see Figure 4). In the stayer sample, average (rounded to the nearest integer) GE, OE, Abstract TE, Routine TE, and 1 year, respectively. A stayer with these characteristics will make about 33.8% more (significant at 10%) more than an immigrant without any work experience (see Figure 4). In the stayer sample, average (see Figure 4). In the switcher sample, average (rounded to nearest integer) GE, OE, Abstract TE, Routine TE, and Yaer, 0 years, 0 years, 1 year, 0 years, 0 years, 1 year, 0 years, 1 years, 0 y

Figure 4 Returns to source country experience for the average immigrant – two-digit level. Returns for "All workers" are significant at the 1% level, returns for "Stayers" are significant at the 10% level, and returns for the Switchers sample are not significant at any of the conventional levels.



and 1 year, respectively. A switcher with these characteristics will make about 9.2% more (not significant) more than immigrant without any work experience (see Figure 4).

As we mentioned earlier, we also estimated the wage equation with the five-task specification. The results for GE and OE are similar to those reported above with the three-task specification. Results also indicate that Analytical (NRCA) TE is the only one that is consistently significant across specifications and samples. This is consistent with the results of threetask specification and implies that returns to Abstract TE are mostly coming from Analytical (NRCA) TE but not Interpersonal (NRCI) TE. The detailed results are presented in Tables S2–S7 in Supplementary Materials.

#### 4.3 Robustness

#### 4.3.1 Controlling for source country wage

One concern with OLS estimates is that they do not account for unobserved heterogeneity. As previously discussed, immigrants with more OE may have unobservable differences relative to individuals with less OE. The positive and significant returns to OE and TE reported from the OLS wage regression may reflect underlying ability differentials. To address this, in addition to observable characteristics, we control for an immigrant worker's wage in the first job in his/ her source country. An immigrant's source country wage may reflect unobservables (such as ability and motivation) not captured in the set of controls used in the analysis. In the NIS data, source country wages are available at two points in time. The first is at the time when an immigrant started working in their source country and the second is in his/her last source country job before immigrating to the U.S. We use the first source country job wage for two reasons: (1) the GE, OE, and TE at the time of his/her first source country job is zero; therefore, the wage reflects the effects of education and unobservables, but not the effects of GE, OE, and TE; (2) it may best reflect the unobservable traits that are easily observed by employers and, therefore, best reflects the unobservables affecting his//her wage at the first U.S. job. In our sample, the first source country job wage is available for 1693 immigrants out of a total of 2459 immigrants. Estimates for all workers suggest that the returns to GE are statistically insignificant, but the returns to OE and Abstract TE are positive and significant.

Panels (a) and (b) of Figure 5 show the results. Panel (a) presents the results from the wage regression based on the sample of immigrants for whom source country wage is available (1693 immigrants), but it does not control for source country wage. Panel (b) shows the results when we add source country wage as an additional control. Panels (a) and (b) suggest that an immigrant with 10 years of GE, but no OE or TE, would earn 0.5% (1.7%) more than an immigrant without any work experience. Neither is statistically significant. An immigrant with 10 years of GE, 3 years of OE, and no TE would earn 17.2% (17.9%) more than an immigrant without any work experience in Panels (a) and (b). Both estimates are significant at 5%. An immigrant with 10 years of GE, 3 years of OE, and 3 years of Abstract TE would earn 28.0% (28.4%) more than an immigrant without any work experience in Panels (a) and (b). Both estimates are significant at 1%.

An alternative way to illustrate returns to work experience is to show how much more an immigrant with average work experience would earn compared to an immigrant without any

Figure 5 Returns to experience: source country wage sample – two-digit level. Note: (a) All immigrant workers. (b) All immigrant workers with source country wage control. (c) Returns for the average immigrant.



*Note:* \*0.1, \*\* 0.05, and \*\* 0.01 denote significance levels under each bar in panels (a) and (b). In panel (c), estimates are significant at the 0.01 level.

work experience. Average (rounded to nearest integer) GE, TE, Abstract TE, Routine TE, and Manual TE in this sample is 11, 3, 2, 0, and 1 year, respectively. Panel (c) of Figure 5 suggests that when we do not control for source country wage, the total return to this work experience portfolio is 24.1%, and when we do control for source country wage, it is 24.9% (see Panel (c) of Figure 5). The results presented here suggest that unobserved heterogeneity may not have a substantial effect on the estimated returns to experience.

**Figure 6** Joint returns to experience, excluding employment principals – two-digit level. Return to 10 years of GE and 0 years of OE and TE are not statistically significant at any of the conventional levels for the "All immigrants" and "Switchers" samples. Return to 10 years of GE and 0 years of OE and TE is significant at the 5% level for the "Stayers" sample. Returns to 10 years of GE, 3 years of OE, and no TE are significant at the 1% level for "All immigrants". Return to 10 years of GE, 3 years of OE, and 0 years of TE is not significant at any of the conventional levels for the "Stayers" and "Switchers" samples. Return to 10 years of GE, 3 years of OE, and 3 years of Abstract TE is significant at the 1% level for the "All immigrants" and "Switchers" samples. Returns to 10 years of GE, 3 years of OE, and 3 years of Abstract TE are not significant at any of the conventional significance levels for the "Stayers" sample.



#### 4.3.2 Robustness with respect to sample criteria

We discussed in the "Data" section that our sample has higher average education and wage compared to employed immigrants in the CPS. Therefore, as a robustness check, we exclude employment principals from our sample. In the results, subsample average education is still higher compared to that in the CPS, but the average wage is lower. We reestimate the returns to source country work experience using this subsample. First, kernel-weighted local polynomial regressions (see Supplementary Figure S1) suggest that this subsample produces results that are similar to the full sample. Regression estimates with the full set of controls also produces congruent results. Figure 6 shows that an immigrant in this subsample with 10 years of GE, but no OE or TE, would earn 8% more (not significant) than an immigrant without any work experience. An immigrant with 10 years of GE, 3 years of OE, and no TE would earn 28% more (significant at 5%) than an immigrant without any work experience. An immigrant without any work experience. Therefore, the positive and significant at 1%) than an immigrant without any work experience. Therefore, the positive and significant return to OE and Abstract TE is robust to exclusion of employment principals from the sample.

## 4.4 Heterogeneity in the returns to experience

In this section, we explore the heterogeneity in the returns to work experience in two dimensions. First, we estimate the returns to experience by skill level (high vs. low). The second dimension of heterogeneity we explore is by source country income (high vs. low).

#### 4.4.1 High-skilled vs. low-skilled immigrants

Farber and Gibbons (1996) have shown that the return to GE is higher for more-educated individuals. Dustmann and Meghir (2005) show similar results for OE. Kwon and Milgrom (2014) have shown that the value of occupation-specific human capital is higher in occupations requiring higher levels of education. However, these papers focus on native populations. We check whether a similar result holds for source country work experience of immigrants. Immigrants may face occupational downgrading after immigrating to the host country (Bauer and Zimmerman, 1999; Akresh, 2008; Chiswick and Miller, 2008; Imai et al., 2017). There is no consensus on the relationship between immigrants' skills and occupational downgrading. In our sample, occupation switch rate varies between 65.6% and 84.7% (depending on occupational granularity) for immigrants without a college degree but is between 35.2% and 70.5% for those with a college degree or more. To determine whether the return to work experience varies with the level of educational attainment of the immigrants, we estimate separate regressions for high-skilled (those with a college degree) and low-skilled (those without a college degree) workers.

Panels (a) and (b) in Figure 7 present the cumulative returns to GE, OE, and TE for lowskilled (without a college degree) workers and high-skilled (with college degree) workers, respectively. In our sample, 1229 (49.96%) immigrants have a college degree, and 1231 (50.04%)





Notes: \*0.1, \*\* 0.05, and \*\* 0.01 denote the significance level. Returns shown are for all workers in each subsample.

immigrants do not have a college degree. The return to 10 years of GE is 6.9% for low-skilled workers and 3.7% for high-skilled workers. If a low-skilled immigrant has 10 years of GE and 3 years of OE, he/she will earn 13.3% (not significant) more than a comparable low-skilled immigrant without any work experience. On the other hand, if a high-skilled immigrant has 10 years of GE and 3 years of OE, he/she will earn 22.1% (significant at 10%) more than a comparable low-skilled immigrant without any work experience. If a low-skilled immigrant has 10 years of GE, 3 years of OE, and 3 years of Abstract TE, he/she will earn 27.5% (significant at 1%) more than a comparable low-skilled immigrant without any work experience. On the other hand, if a high- skilled immigrant has 10 years of GE, 3 years of OE, and 3 years of Abstract TE, he/she will earn 30.4% (significant at 1%) more than a comparable low-skilled immigrant without any work experience. The cumulative return estimates to experience suggest that lowand high-skilled immigrants with the same amount of GE, OE, and TE receive similar returns. However, the average experience is different across these two groups. Average GE, OE, Abstract TE, Routine TE, and Manual TE for low-skilled workers are 13, 2, 1, 1, and 2 years, respectively. The corresponding numbers for high-skilled workers are 11, 3, 4, 0, and 0 years. Given these, a low-skilled immigrant with average work experience profile earns 19.8% (high-skilled: 32.2%) more than a low-skilled (high-skilled) immigrant without any work experience (see Figure 8).

#### 4.4.2 High-income vs. low-income source countries

Source country heterogeneity may be another important source of worker unobservables (Adsera and Chiswick, 2007). Aydemir and Skuterud (2005) have found that immigrants from "nontraditional" (mostly developing countries) who immigrate to Canada get a lower return to their source country GE. While the analysis above uses source country fixed effects to identify parameters from within-country variation, the estimated parameters still represent the average over all the source countries. To check whether returns to GE, OE, and TE vary by source country characteristics (namely, income), we group the source countries into two categories: low-income and high-income countries, as described in Section 3.

Panels (c) and (d) of Figure 7 show the results for immigrants originating in high-income and low-income countries, respectively. In our sample, 584 (23.74%) immigrants are from high-income countries, and 1876 (76.26%) immigrants are from low-income countries. The



**Figure 8** Average returns to experience – worker heterogeneity (two-digit level).

Notes: All returns are significant at the 1% level.

return to 10 years of GE is 6.3% for immigrants from low-income countries and 18.2% for immigrants from high-income countries. Neither is statistically significant. If an immigrant from a low-income country has 10 years of GE and 3 years of OE, he/she will earn 23.3% (significant at 5%) more than a comparable immigrant without any work experience. On the other hand, if an immigrant from a high-income country has 10 years of GE and 3 years of OE, then he/she will earn 30.8% (significant at 10%) more than a comparable immigrant without any work experience. If an immigrant from a low-income country has 10 years of GE, 3 years of OE, and 3 years of Abstract TE, then he/she will earn 32.8% (significant at 1%) more than a comparable immigrant without any work experience. On the other hand, if an immigrant from a high-income country has 10 years of GE, 3 years of OE, and 3 years of Abstract TE, he/she will earn 45.3% (significant at 1%) more than a comparable low-skilled immigrant without any work experience. This suggests that returns to work experience are higher for immigrants from high-income countries compared to those from low-income countries. Average GE, OE, Abstract TE, Routine TE, and Manual TE for immigrants from low-income countries are 12, 2, 2, 1, and 1 years, respectively. The corresponding numbers for high-skilled workers are 13, 3, 3, 1, and 2 years. Given these, an immigrant from a low-income (high-income) country with an average work experience profile earns 23.7% (39.2%) more than an immigrant from a low-income (high-income) country without any work experience (see Figure 8).

## 5 Discussion and conclusion

This article sheds new light on the portability of human capital using data from new immigrants to the U.S. Using wages earned by immigrants in their first U.S. job, we show that human capital may be more portable across countries than previously reported. In particular, we show that even though the return to source country general experience is small or even negligible, returns to occupation-specific experience accumulated in the source country is statistically and economically significant. Moreover, we find that task-specific human capital is an important component, and human capital associated with Abstract tasks is transferable across countries. We find consistent evidence that Abstract TE, especially NRCA TE acquired in source countries of the immigrants has significant wage returns in the U.S. However, we find that while analytical TE is transferable, interpersonal TE is not. The difference in transferability between the two high-skilled associated measures of TE may stem from the fact that NRCI skills are more closely tied to English communication skills than are those associated with performing NRCA tasks. This evidence of little-to-no transferability of interpersonal task-specific human capital may also explain why a large number of immigrants sort into non-NRCI-intensive occupations after immigrating to a host country (Peri and Sparber, 2009). We also find that routine and manual task-specific work experiences acquired in the source countries of the immigrants do not bring significant wage return in the U.S. labor market. We are not aware of any previous research that has estimated returns to source country OE and TE for immigrants. Our primary results are robust to changes in specification (such as controlling for unobserved heterogeneity by including source country wage as a control) and sample criteria (such as excluding employment principals from our sample).

We should note that, in the context of TE, we are approximating a source country's occupation technologies using U.S. (ONET) data. Although there is some evidence that the rankings of tasks are similar across countries (Dicarlo et al., 2016), the specificity of technologies and, therefore, their rankings may differ at different levels of occupational coding. It is plausible that at the one-digit level, occupations are too broadly defined, and, given such a broad definition, composite technologies of occupations at this level do vary across countries. Similarly, at the three-digit level, occupation technologies may be too specific, and approximating tasks across countries based on rank may not be the most consistent. Further exploration of the limiting factors discussed here may be addressed by researchers in the future.

#### References

- Acemoglu, D.; D. H. Autor (2011): Skills, Tasks and Technologies: Implications for Employment and Earnings, in: Ashenfelter, O.; D. E. Card (eds.), Handbook of Labor Economics. Vol. 4B. Amsterdam: Elsevier, 1043-1171.
- Adsera, A.; B. R. Chiswick (2007): Are There Gender and Country of Origin Differences in Immigrant Labor Market Outcomes Across European Destinations? Journal of Population Economics 20(3), 495-526.
- Akresh, I. R. (2008): Occupational Trajectories of Legal US Immigrants: Downgrading and Recovery. Population and Development Review 34(3), 435-456.
- Altonji, J. G.; R. A. Shakotko (1987): Do Wages Rise with Job Seniority? The Review of Economic Studies 54(3), 437.
- Autor, D. H. (2013): The Task Approach to Labor Markets: An Overview. Journal of Labour Market Research 46(3), 1-15.
- Autor, D. H.; M. J. Handel (2013): Putting Tasks to the Test: Human Capital, Job, Tasks, and Wages. Journal of Labor Economics 31(2), 559-596.
- Autor, D. H.; F. Levy; R. J. Murnane (2003): The Skill Content of Recent Technological Change: An Empirical Exploration. The Quarterly Journal of Economics 118(4), 1279-1333.
- Aydemir, A.; M. Skuterud (2005): Explaining the Deteriorating Entry Earnings of Canada's Immigrant Cohorts, 1966-2000. Canadian Journal of Economics 38(2), 641-672.
- Basilio, L.; T. K. Bauer; A. Kramer (2017): Transferability of Human Capital and Immigrant Assimilation: An Analysis for Germany. Labour 31(3), 245-264.
- Bauer, T.; K. F. Zimmerman (1999): Assessment of Possible Migration Pressure and its Labour Market Impact Following EU Enlargement to Central and Eastern Europe, IZA Research Report No. 3.
- Beenstock, M.; B. R. Chiswick; A. Paltiel (2009): Testing the Immigrant Assimilation Hypothesis with Longitudinal Data. Review of Economics of the Household 8(1), 7-27.
- Bratsberg, B., Jr.; J. F. Ragan (2002): The Impact of Host-Country Schooling on Earnings: A Study of Male Immigrants in the United States. The Journal of Human Resources 37(1), 63-105.
- Chiswick, B. R.; Y. L. Lee; P. W. Miller (2005): Immigrant Earnings: A Longitudinal Analysis. Review of Income and Wealth 51(4), 485-503.
- Chiswick, B. R.; P. W. Miller (2008): Occupational Attainment and Immigrant Economic Progress in Australia. Economic Record 84, S45-S56.
- Cohen-Goldner, S.; Z. Eckstein (2008): Labor Mobility of Immigrants: Training, Experience, Language, and Opportunities. International Economic Review 49(3), 837-872.
- Cortes, G. M.; G. Gallipoli (2017): The Costs of Occupational Mobility: An Aggregate Analysis. Journal of the European Economics Association, jvx006.
- Dicarlo, E.; S. L. Bello; S. Monroy-Taborda, A. M. Oviedo; M. L. Sanchez-Puerta; I. Santos (2016): The Skill Content of Occupations across Low and Middle Income Countries: Evidence from Harmonized Data. IZA Discussion Paper no. 10224(September), IZA, Bonn.
- Dustmann, C.; C. Meghir (2005): Wages, Experience and Seniority. The Review of Economic Studies 72(1), 77-108.
- Farber, H. S.; R. Gibbons (1996): Learning and Wage Dynamics. The Quarterly Journal of Economics 111(4), 1007-1047.
- Ferrer, A.; D. A. Green; W. C. Riddell (2006): The Effect of Literacy on Immigrant Earnings. Journal of Human Resources 41(2), 380-410.
- Ferrer, A.; W. C. Riddell (2008): Education, Credentials, and Immigrant Earnings. Canadian Journal of Economics 41(1), 186-216.
- Friedberg, R. M. (2000): You Can't Take It with You? Immigrant Assimilation and the Portability of Human Capital. Journal of Labor Economics 18(2), 221-251.
- Gathmann, C.; U. Schönberg (2010): How General Is Human Capital? A Task-Based Approach. Journal of Labor Economics 28(1), 1-49.

Gibbons, R.; M. Waldman (2004): Task-Specific Human Capital. The American Economic Review 94(2), 203-207. Green, D. A. (2009): New Academics' Perceptions of the Language of Teaching and Learning: Identifying and

- Overcoming Linguistic Barriers. International Journal for Academic Development 14(1), 33-45. Green, D. A.; C. Worswick (2002): Earnings of Immigrant Men in Canada: The Roles of Labour Market Entry
- Effects and Returns to Foreign Experience. Study prepared for Citizenship and Immigration Canada. Grieco, E. M.; E. N. Trevelyan (2010, October): Places of Birth of the Foreign-Born Population: 2009. American Community Survey Briefs.
- Guillermina, J.; D. S.; Massey, M. R. Rosenzweig J. P. Smith (2006): The New Immigrant Survey 2003 Round 1 (NIS-2003-1) Public Release Data. Retrieved 2016. Funded by NIH HD33843, NSF, USCIS, ASPE & Pew. http://nis.princeton.edu.
- Imai, S.; D. Stacy; C. Warman (2017): From Engineer to Taxi Driver? Language Proficiency and the Occupational Skills of Immigrants, Working Paper.
- Kambourov, G.; I. Manovskii (2009): Occupational Specificity of Human Capital. International Economic Review 50(1), 63-115.
- Kambourov, G.; I. Manovskii (2012): A Cautionary Note on Using (March) Current Population Survey and Panel Study of Income Dynamics Data to Study Worker Mobility. Macroeconomic Dynamics 17(01), 172-194.
- Kossoudji, S. A. (1989): Immigrant Worker Assimilation: Is It a Labor Market Phenomenon? The Journal of Human Resources 24(3), 494-527.
- Kwon, I.; E. M. M. Milgrom, (2014): The Significance of Firm and Occupation Specific Human Capital for Hiring and Promotions. Labour Economics 31, 162-173.
- Mellow, W.; H. Sider (1983): Accuracy of Response in Labor Market Surveys: Evidence and Implications. Journal of Labor Economics 1(4), 331-344.
- Mincer, J. A. (1974): The Human Capital Earnings Function. Schooling, Experience, and Earnings. NBER 83-96.
- Moscarini, G.; K. Thomsson (2007): Occupational and Job Mobility in the US. Scandinavian Journal of Economics 109(4), 807-836.
- Murphy, K. M.; F. Welch (1990): Empirical Age-Earnings Profiles. Journal of Labor Economics 8(2), 202-229.
- Neal, D. (1995): Industry-specific Human Capital: Evidence from Displace Workers. Journal of Labor Economics 13(4), 653-677.
- Nielson, C. P. (2011): Immigrant Over-Education: Evidence from Denmark. Journal of Population Economics 24(2), 499-520.
- O\*NET. (2003): Resource Center, National Center for O\*NET Development, www.onetcenter.org/bd\_releases. html. Accessed 2016.
- Parent, D. (2000): Industry-specific Capital and the Wage Profile: Evidence from the National Longitudinal Survey of Youths and the Panel Study of Income Dynamics. Journal of Labor Economics 18(2), 306-321.
- Peri, G.; C. Sparber (2009): Task Specialization, Immigration, and Wages. American Economic Journal: Applied Economics 1(3), 135-169.
- Reinhold, S.; K. Thom (2013): Migration Experience and Earnings in the Mexican Labor Market. The Journal of Human Resources 48(3), 768-820.
- Schaafsma, J.; A. Sweetman (2001): Immigrant Earnings: Age at Immigration Matters. Canadian Journal of Economics 34(4), 1066-1099.
- Schulz, E.; S. Chowdhury; D. V. de Voort (2013): Firm Productivity Moderated Link Between Human Capital and Compensation: The Significance of Task-Specific Human Capital. Human Resource Management 52(3), 423-439.
- Sullivan, P. (2010): Empirical Evidence on Occupation and Industry Specific Human Capital. Labour Economics 17(3), 567-580.
- Topel, R. (1991): Specific Capital, Mobility, and Wages: Wages Rise with Job Security. Journal of Political Economy 99(1), 145-176.
- Williams, N. (2009): Seniority, Experience, and Wages in the UK. Labour Economics 16(3), 272-283.
- Worswick, C. (2003): Thinking the Unthinkable: The Immigration Myth Exposed. Journal of Economic Literature 41(3), 923-924.

# **Supplementary Materials**





Parameters	Alli	mmigrant work	(ers		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.0291	0.0478	0.0421	0.0658	0.0507	0.1632	0.0449	0.0254	0.0114
	(0.0368)	(0.0373)	(0.0368)	(0.1165)	(0.1336)	(0.1556)	(0.0429)	(0.0401)	(0.0386)
5 Years of GE	0.0402	0.0654	0.0593	0.0838	0.0617	0.2297	0.0645	0.0367	0.0185
	(0.0502)	(0.0510)	(0.0504)	(0.1573)	(0.1806)	(0.2080)	(0.0588)	(0.0548)	(0.0528)
10 Years of GE	0.0470	0.0744	0.0718	0.0707	0.0395	0.3011	0.0866	0.0490	0.0283
	(0.0606)	(0.0621)	(0.0614)	(0.1845)	(0.2148)	(0.2457)	(0.0715)	(0.0667)	(0.0641)
3 Years of OE	0.1526***	0.1699***	0.2024***	0.1043	0.2623**	0.1176	0.1109	0.0891	$0.1184^{*}$
	(0.0250)	(0.0306)	(0.0337)	(0.1199)	(0.1255)	(0.1449)	(0.0754)	(0.0577)	(0.0660)
5 Years of OE	0.2033***	0.2251***	0.2719***	0.1387	0.3516**	0.1596	0.0642	0.1056	0.2672**
	(0.0316)	(0.0374)	(0.0413)	(0.1575)	(0.1640)	(0.1868)	(0.0899)	(0.0752)	(0.1224)
10 Years of OE	0.2262***	0.2381***	0.3059***	0.1582	0.3941**	0.1910	-0.0871	0.0513	0.1871
	(0.0390)	(0.0414)	(0.0458)	(0.1704)	(0.1758)	(0.1933)	(0.1314)	(0.1341)	(0.1827)
3 Years of Abstract	0.1239***	0.0997***	0.1066***	0.0911***	0.0588*	0.0443	0.1272***	0.1292***	0.1541***
	(0.0196)	(0.0186)	(0.0186)	(0.0267)	(0.0304)	(0.0384)	(0.0428)	(0.0295)	(0.0300)
5 Years of Abstract	0.2017***	0.1623***	0.1738***	0.1376***	0.0846*	0.0609	0.2065***	0.2086***	0.2473***
	(0.0311)	(0.0295)	(0.0295)	(0.0450)	(0.0500)	(0.0628)	(0.0658)	(0.0466)	(0.0470)
10 Years of Abstract	0.3614***	0.2898***	0.3104***	0.1911**	0.0944	0.0508	0.3653***	0.3588***	0.4141***
	(0.0505)	(0.0478)	(0.0484)	(0.0955)	(0.1067)	(0.1334)	(0.0929)	(0.0736)	(0.0715)
3 Years of Routine	-0.0317	-0.0380	-0.0204	-0.0223	0.0281	0.0011	-0.0452	-0.0384	-0.0079
	(0.0252)	(0.0292)	(0.0216)	(0.0294)	(0.0593)	(0.0864)	(0.0725)	(0.0399)	(0.0247)
									(Continued)

Parameters	Allir	nmigrant worl	kers		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 Years of Routine	-0.0542	-0.0593	-0.0314	-0.0287	0.0629	0.0047	-0.0247	-0.0574	-0.0116
	(0.0400)	(0.0460)	(0.0347)	(0.0465)	(0.0934)	(0.1362)	(0.1029)	(0.0624)	(0.0395)
10 Years of Routine	-0.1124*	-0.0866	-0.0445	-0.0124	0.2059	0.0217	0.2636	-0.0636	-0.0121
	(0.0638)	(0.0715)	(0.0579)	(0.0799)	(0.1498)	(0.2213)	(0.2620)	(0.0927)	(0.0655)
3 Years of Manual	-0.0273	-0.0128	-0.0295*	-0.0262	-0.0157	-0.0371	-0.0281	-0.0040	-0.0209
	(0.0194)	(0.0185)	(0.0175)	(0.0336)	(0.0468)	(0.0553)	(0.0357)	(0.0210)	(0.0184)
5 Years of Manual	-0.0437	-0.0225	-0.0490*	-0.0389	-0.0319	-0.0583	-0.0411	-0.0084	-0.0355
	(0.0313)	(0.0296)	(0.0280)	(0.0563)	(0.0765)	(0.0929)	(0.0552)	(0.0337)	(0.0296)
10 Years of Manual	-0.0744	-0.0522	-0.0935**	-0.0458	-0.0963	-0.0953	-0.0426	-0.0269	-0.0727
	(0.0533)	(0.0495)	(0.0465)	(0.1135)	(0.1459)	(0.1949)	(0.0797)	(0.0570)	(0.0503)
Z	2,459	2,459	2,459	1,006	694	551	1,453	1,765	1,908
Note: Standard errors an	e in parentheses								
*0.10, **0.05, and ***0.01	denote signific	ance levels.							

Table S1 Continued

	ŝ
	6
	≶
	le
	a
	Ŭ
)	3
5	۲
j	÷
	50
	5
)	d)
5	ž
2	2
	ē
	Ō
)	-
5	0.
)	0
5	*
	*
,	σ
5	Ē
	10
5	S
	o.
5	Õ
5	*
	ć
)	2

Parameters	Allin	nmigrant worke	ers		Stayers		S	Switchers	
I	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.0398	0.0448	0.0443	0.1139	-0.0043	0.1525	0.0575	0.0319	0.0245
	(0.0365)	(0.0375)	(0.0373)	(0.1192)	(0.1380)	(0.1543)	(0.0424)	(0.0404)	(0.0394)
5 Years of GE	0.0568	0.0654	0.0650	0.1471	0.0042	0.2298	0.0841	0.0485	0.0386
	(0.0498)	(0.0514)	(0.0510)	(0.1605)	(0.1849)	(0.2069)	(0.0582)	(0.0554)	(0.0539)
10 Years of GE	0.0763	0.0901	0.0885	0.1346	0.0432	0.3502	0.1216*	0.0728	0.0590
	(0.0604)	(0.0628)	(0.0620)	(0.1866)	(0.2133)	(0.2405)	(0.0718)	(0.0678)	(0.0653)
3 Years of OE	0.1492***	0.1678***	0.1872***	0.0394	0.2944**	0.1157	0.5501***	0.0519	0.0719
	(0.0261)	(0.0319)	(0.0349)	(0.1216)	(0.1288)	(0.1437)	(0.1616)	(0.0675)	(0.0789)
5 Years of OE	0.1981***	0.2232***	0.2539***	0.0558	0.3836**	0.1505	1.4284**	0.0544	0.1610
	(0.0335)	(0.0392)	(0.0431)	(0.1598)	(0.1675)	(0.1852)	(0.5558)	(0.0844)	(0.1443)
10 Years of OE	0.2163***	0.2388***	0.2897***	0.0787	0.3852**	0.1610	7.5671**	0.0246	0.1821
	(0.0419)	(0.0427)	(0.0479)	(0.1729)	(0.1768)	(0.1905)	(3.7792)	(0.1355)	(0.2131)
3 years of NRCA	0.1725***	0.1693***	0.1305***	0.1306***	0.0892*	0.0376	0.5014***	0.5412***	0.2400***
	(0.0301)	(0.0355)	(0.0239)	(0.0346)	(0.0456)	(0.0406)	(0.1671)	(0.1274)	(0690.0)
5 years of NRCA	0.2708***	0.2690***	0.2103***	0.2026***	0.1195	0.0438	0.1480	0.7628***	0.3798***
	(0.0476)	(0.0552)	(0.0379)	(0.0593)	(0.0734)	(0.0726)	(0.2060)	(0.1839)	(0.1065)
10 years of NRCA	0.4272***	0.4366***	0.3582***	0.3089**	0.0711	-0.0093	-5.8783**	0.5968**	0.5885***
	(0.0775)	(0.0832)	(0.0628)	(0.1448)	(0.1445)	(0.1874)	(2.8815)	(0.2753)	(0.1538)
3 years of NRCI	0.0708**	-0.0106	-0.0377	0.0697**	-0.0733	$-0.1589^{**}$	-0.1333	0.0309	-0.0246
	(0.0291)	(0.0333)	(0.0365)	(0.0341)	(0.0628)	(0.0731)	(0.1665)	(0.0381)	(0.0500)
									(Continued)

Table S2 Continued									
Parameters	Alli	mmigrant work	ers		Stayers			Switchers	
I	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 years of NRCI	0.1144**	-0.0153	-0.0583	0.0991*	-0.1207	-0.2650**	-0.7963**	0.0468	-0.0372
	(0.0457)	(0.0520)	(0.0581)	(0.0552)	(0.1011)	(0.1216)	(0.3748)	(0.0592)	(0.0799)
10 years of NRCI	0.1985***	-0.0089	-0.0843	0.0952	-0.2041	-0.4890*	$-6.1920^{**}$	0.0625	-0.0477
	(0.0719)	(0.0782)	(0.0946)	(0.1031)	(0.1819)	(0.2511)	(3.0430)	(0.0870)	(0.1333)
3 years of RCOG	-0.0404	0.0558*	0.0049	-0.0343	0.0524	0.0123	-0.2918***	0.0767	0.0228
	(0.0263)	(0.0290)	(0.0287)	(0.0297)	(0.0466)	(1660.0)	(0.1039)	(0.0560)	(0.0339)
5 years of RCOG	-0.0693*	0.0866*	0.0075	-0.0513	0.0888	0.0204	-0.8822***	0.1237	0.0365
	(0.0415)	(0.0456)	(0.0458)	(0.0467)	(0.0769)	(0.1564)	(0.3146)	(0.0850)	(0.0538)
10 years of RCOG	$-0.1430^{**}$	0.1274*	0.0100	-0.0671	0.1708	0.0265	-5.5867**	0.2133*	0.0589
	(0.0656)	(0.0714)	(0.0745)	(0.0789)	(0.1539)	(0.2615)	(2.7181)	(0.1112)	(0.0869)
3 years of RMAN	-0.0713**	-0.1062***	-0.1233***	-0.0249	-0.1196	-0.1748	-0.3068***	-0.0993**	-0.1217**
	(0.0361)	(0.0361)	(0.0476)	(0.0537)	(0.0770)	(0.1368)	(0.0990)	(0.0491)	(0.0522)
5 years of RMAN	$-0.1086^{*}$	-0.1617***	-0.1869**	-0.0212	-0.1583	-0.2506	-0.9394***	$-0.1500^{**}$	$-0.1830^{**}$
	(0.0572)	(0.0563)	(0.0734)	(0.0876)	(0.1234)	(0.2118)	(0.3196)	(0.0746)	(0.0793)
10 years of RMAN	-0.1418	-0.2143**	-0.2329**	0.0829	-0.0590	-0.2258	-5.8974**	$-0.1916^{*}$	-0.2180*
	(0.0913)	(0.0850)	(0.1069)	(0.1618)	(0.2164)	(0.3226)	(2.8076)	(0.1000)	(0.1151)
3 years of NRMN	-0.0646**	-0.0167	-0.0482**	-0.0598	0.0143	-0.0204	-0.2996***	-0.0085	-0.0368
	(0.0327)	(0.0220)	(0.0219)	(0.0469)	(0.0513)	(0.0656)	(0.1016)	(0.0260)	(0.0251)
5 years of NRMN	-0.0978	-0.0273	-0.0771**	-0.0921	0.0182	-0.0315	-0.9177***	-0.0142	-0.0594
	(0.0528)	(0.0352)	(0.0349)	(0.0803)	(0.0814)	(0.1089)	(0.3269)	(0.0417)	(0.0397)
10 years of NRMN	-0.1335	-0.0532	$-0.1314^{**}$	-0.1293	-0.0043	-0.0566	-5.7884**	-0.0302	-0.1055*
	(0.0915)	(0.0588)	(0.0577)	(0.1834)	(0.1390)	(0.2267)	(2.8214)	(0.0703)	(0.0637)
N	2,459	2,459	2,459	1,006	694	551	1,453	1,765	1,908
<i>Note</i> : Standard errors ar * 0.10, ** 0.05, and *** 0.	e in parenthese 01 denote signif	s. ficance levels.							

				•					
Parameters	All im	migrant worker	S		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	-0.0195	-0.0124	-0.0195	0.1548	0.0692	0.4922*	0.0610	0.0105	0.0030
	(0.0466)	(0.0472)	(0.0475)	(0.1875)	(0.2111)	(0.2893)	(0.0570)	(0.0512)	(0.0490)
5 Years of GE	-0.0150	-0.0079	-0.0147	0.2055	0.1009	0.6256*	0.0930	0.0215	0.0145
	(0.0624)	(0.0635)	(0.0637)	(0.2465)	(0.2692)	(0.3569)	(0.0768)	(0.0692)	(0.0661)
10 Years of GE	0.0223	0.0218	0.0201	0.2150	0.1635	0.6601*	0.1506*	0.0488	0.0483
	(0.0725)	(0.0740)	(0.0739)	(0.2667)	(0.2736)	(0.3366)	(0.0908)	(0.0817)	(0.0779)
3 Years of OE	0.1700***	0.1568***	0.1821***	-0.0341	0.0601	-0.4223	0.4687***	0.0303	0.0282
	(0.0317)	(0.0388)	(0.0424)	(0.1853)	(0.2104)	(0.2668)	(0.1773)	(0.0827)	(0.1008)
5 Years of OE	0.2234***	0.2191***	0.2476***	-0.0315	0.1028	-0.4835	$1.0855^{*}$	-0.0674	0.0866
	(0.0399)	(0.0449)	(0.0487)	(0.2398)	(0.2628)	(0.3216)	(0.5809)	(0.1763)	(0.2094)
10 Years of OE	0.2417***	0.2639***	0.2861***	0.0177	0.1623	-0.3262	4.5032	-0.0570	-0.0487
	(0.0520)	(0.0518)	(0.0556)	(0.2482)	(0.2499)	(0.2772)	(4.0280)	(0.3098)	(0.3089)
3 years of NRCA	0.1722***	0.2147***	0.1882***	0.1136**	0.0826	-0.0102	0.5413**	0.5159***	0.4403***
	(0.0402)	(0.0418)	(0.0386)	(0.0551)	(0.0591)	(0.0554)	(0.2176)	(0.1154)	(0.1195)
5 years of NRCA	0.2670***	0.3316***	0.2905***	0.1859*	0.0860	-0.0858	0.3632**	0.7872***	0.6519***
	(0.0612)	(0.0627)	(0.0566)	(0.1126)	(0.1038)	(0.0958)	(0.1756)	(0.1675)	(0.1678)
10 years of NRCA	0.3993***	0.4748***	0.4102***	0.3450	-0.1110	$-0.5139^{**}$	-3.7969	1.0420***	0.7375***
	(0.0902)	(0.0862)	(0.0734)	(0.3805)	(0.2727)	(0.2445)	(3.0452)	(0.2512)	(0.1912)
3 years of NRCI	0.0703*	-0.0159	-0.0646	0.0570	-0.1298	-0.1282	-0.0536	0.0248	-0.0847
	(0.0413)	(0.0424)	(0.0435)	(0.0481)	(0.0892)	(0.1335)	(0.1866)	(0.0483)	(0.0526)
									(Continued)

Table S3 Continued									
Parameters	All im	migrant worke	ers		Stayers			Switchers	
-	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 years of NRCI	0.1142*	-0.0231	-0.0984	0.0939	-0.1964	-0.2305	-0.8242**	0.0373	-0.1310
	(0.0629)	(0.0662)	(0.0688)	(0.0776)	(0.1427)	(0.2044)	(0.3591)	(0.0754)	(0.0835)
10 years of NRCI	0.2047**	-0.0119	-0.1251	0.1756	-0.2299	-0.5080	15.9488***	0.0514	-0.1809
	(0.0865)	(0.0994)	(0.1081)	(0.1595)	(0.2893)	(0.3149)	(5.0011)	(0.1122)	(0.1354)
3 years of RCOG	-0.0689**	0.0629	-0.0102	-0.0581	0.0015	-0.0134	-0.2584**	0.1217*	0.0005
	(0.0318)	(0.0408)	(0.0300)	(0.0425)	(0.0633)	(0.1919)	(0.1222)	(0.0705)	(0.0329)
5 years of RCOG	$-0.1164^{**}$	0.1019	-0.0154	-0.0880	0.0123	0.0314	-0.7541**	0.2015*	0.0014
	(0.0501)	(0.0631)	(0.0480)	(0.0737)	(0.1063)	(0.2811)	(0.3393)	(0.1069)	(0.0526)
10 years of RCOG	-0.2326***	0.1798*	-0.0191	-0.1195	0.0798	0.3691	-3.8501	0.3832***	0.0071
	(0.0792)	(0.0937)	(0.0800)	(0.1832)	(0.2485)	(0.3505)	(2.9752)	(0.1406)	(0.0876)
3 years of RMAN	-0.0437	-0.0712	-0.0618	-0.0184	-0.1597*	-0.1242	-0.2327	-0.0430	-0.0344
	(0.0776)	(0.0475)	(0.0870)	(0.1052)	(0.0887)	(0.1785)	(0.1509)	(0.0633)	(0.1042)
5 years of RMAN	-0.0673	-0.1047	-0.0936	0.0003	-0.2173	-0.1384	-0.6033*	-0.0568	-0.0476
	(0.1193)	(0.0743)	(0.1347)	(0.1669)	(0.1449)	(0.2681)	(0.3569)	(0.0973)	(0.1590)
10 years of RMAN	-0.0935	-0.1166	-0.1152	0.1843	-0.1249	0.1254	-3.3147	-0.0182	-0.0265
	(0.1640)	(0.1122)	(0.1916)	(0.2899)	(0.2959)	(0.3855)	(2.9932)	(0.1346)	(0.2162)
3 years of NRMN	-0.0572	-0.0058	-0.0639*	-0.0786	0.1113	-0.0907	-0.3180***	0.0026	-0.0662*
	(0.0372)	(0.0291)	(0.0369)	(0.0673)	(0.1020)	(0.1091)	(0.1201)	(0.0366)	(0.0355)
5 years of NRMN	-0.0865	-0.0134	-0.1033*	-0.1596	0.1618	-0.1241	-0.7497**	0.0007	$-0.1041^{*}$
	(0.0597)	(0.0463)	(0.0556)	(0.1370)	(0.1617)	(0.1697)	(0.3457)	(0.0587)	(0.0544)
10 years of NRMN	-0.1181	-0.0519	$-0.1805^{**}$	-0.4455	0.1418	-0.0717	-3.5858	-0.0231	-0.1593**
	(0.1025)	(0.0766)	(0.0726)	(0.4207)	(0.3003)	(0.2907)	(3.0164)	(0.0983)	(0.0784)
Z	1,693	1,693	1,693	745	528	436	948	1,165	1,257
Note: Standard errors a	re in parentheses.								

 $^{\ast}$  0.10,  $^{\ast\ast}$  0.05, and  $^{\ast\ast\ast}$  0.01 denote significance levels.

				•					
Parameters	All im	migrant work	ers		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.0321	0.0111	0.0281	0.0204	-0.0784	0.1812	0.0492	0.0050	0.0024
	(0.0424)	(0.0435)	(0.0432)	(0.2204)	(0.2901)	(0.4032)	(0.0552)	(0.0471)	(0.0453)
5 Years of GE	0.0487	0.0226	0.0456	0.0415	-0.1135	0.2176	0.0751	0.0153	0.0106
	(0.0582)	(0.0600)	(0.0596)	(0.2914)	(0.3826)	(0.5436)	(0.0763)	(0.0653)	(0.0627)
10 Years of GE	0.0744	0.0548	0.0805	0.0851	-0.1422	0.1868	0.1250	0.0519	0.0393
	(0.0713)	(0.0745)	(0.0737)	(0.3240)	(0.4218)	(0.6501)	(0.0954)	(0.0823)	(0.0783)
3 Years of OE	0.0021	0.0469	0.0690	-0.0337	0.2275	0.1953	0.1271	-0.0590	-0.0385
	(0.0345)	(0.0446)	(0.0513)	(0.2391)	(0.2964)	(0.4180)	(0.1631)	(0.0867)	(0.1096)
5 Years of OE	-0.0031	0.0633	0.0938	-0.0673	0.2975	0.2536	0.0668	-0.0619	-0.1182
	(0.0435)	(0.0548)	(0.0625)	(0.3101)	(0.3933)	(0.5641)	(0.2445)	(0.0995)	(0.2185)
10 Years of OE	-0.0099	0.0753	0.1122*	-0.1301	0.2918	0.2250	-0.0404	0.0352	0.2721**
	(0.0505)	(0.0560)	(0.0635)	(0.3230)	(0.4352)	(0.6658)	(0.5675)	(0.1542)	(0.1153)
3 Years of NRCA	0.4488***	0.3350**	0.3329***	0.4362***	0.4590***	0.3877***	1.3619	0.0272	-1.8790
	(0.1081)	(0.1662)	(0.0781)	(0.1004)	(0.1700)	(0.1298)	(1.2122)	(0.2084)	(1.8057)
5 Years of NRCA	0.6995***	0.5224**	0.5329***	0.6630***	0.6845**	0.6143***	1.7608	0.0427	-13.5495
	(0.1664)	(0.2519)	(0.1233)	(0.1676)	(0.2638)	(0.2115)	(1.5745)	(0.2720)	(12.1213)
10 Years of NRCA	1.0409***	0.7750**	0.8846***	0.8944**	0.8183*	0.9816**	-1.2174	0.0629	-124.7749
	(0.2348)	(0.3232)	(0.1933)	(0.3570)	(0.4395)	(0.4049)	(1.2194)	(0.2713)	(109.8213)
3 Years of NRCI	0.1920***	0.1447**	0.0464	$0.1659^{*}$	0.0843	0.0650	-0.6430	0.0837	0.0127
	(0.0649)	(0.0650)	(0.0523)	(0.0860)	(0.1811)	(0.1587)	(1.0663)	(0.0640)	(0.0663)
									(Continued)

 Table S4
 OLS estimates of cumulative returns to source country GE, OE, and TE (quartic): low-skilled workers

Parameters	All in	imigrant work	ers		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 Years of NRCI	0.3022***	0.2188**	0.0752	0.2506*	0.1210	0.0573	-4.2488	0.1220	0.0229
	(0.1021)	(0.0982)	(0.0835)	(0.1362)	(0.2850)	(0.2501)	(5.5933)	(0.0973)	(0.1063)
10 Years of NRCI	0.4650***	0.2744**	0.1316	0.3238	0.1206	-0.1886	-18.0374	0.1154	0.0594
	(0.1574)	(0.1293)	(0.1388)	(0.2206)	(0.4585)	(0.4406)	(22.5806)	(0.1301)	(0.1780)
3 Years of RCOG	0.0695**	0.1542**	0.0210	0.0784**	0.2713***	-0.2451	-0.1316	0.1670	0.0292
	(0.0319)	(0.0687)	(0.0368)	(0.0397)	(0.0954)	(0.2045)	(0.1350)	(0.2104)	(0.0437)
5 Years of RCOG	0.1081**	0.2450**	0.0362	0.1357*	0.4236**	-0.3285	-0.1305	0.2679	0.0509
	(0.0505)	(0.1081)	(0.0580)	(6020.0)	(0.1641)	(0.3469)	(0.2233)	(0.2912)	(0.0682)
10 Years of RCOG	0.1597**	0.3911**	0.0799	0.2807	0.6208*	-0.1600	0.3139	0.4407*	0.1167
	(0.0810)	(0.1659)	(0.0902)	(0.1931)	(0.3596)	(0.7545)	(0.6260)	(0.2657)	(0.1020)
3 Years of RMAN	0.0223	-0.0267	-0.0564	0.0023	0.0399	-0.0587	0.0302	-0.0398	-0.0611
	(0.0406)	(0.0402)	(0.0523)	(0.0672)	(0.1097)	(0.2125)	(0.1273)	(0.0538)	(0.0585)
5 Years of RMAN	0.0362	-0.0413	-0.0830	0.0111	0.0981	-0.0493	0.0243	-0.0617	-0.0889
	(0.0644)	(0.0630)	(0.0808)	(0.1113)	(0.1833)	(0.3372)	(0.2105)	(0.0813)	(0.0903)
10 Years of RMAN	0.0610	-0.0606	-0.0836	0.0660	0.3485	0.1781	-0.1116	-0.0878	-0.0823
	(0.1031)	(0.0954)	(0.1190)	(0.2241)	(0.3701)	(0.5942)	(0.5349)	(0.1080)	(0.1336)
3 Years of NRMN	0.0451	0.0393	0.0093	0.0608	0.0528	0.0084	-0.0012	0.0344	0.0230
	(0.0370)	(0.0251)	(0.0225)	(0.0558)	(0.0600)	(0660.0)	(0.1388)	(0.0301)	(0.0250)
5 Years of NRMN	0.0777	0.0599	0.0117	0.1188	0.0850	-0.0344	0.0055	0.0512	0.0317
	(0.0597)	(0.0402)	(0.0360)	(0.1093)	(0660.0)	(0.1636)	(0.2321)	(0.0484)	(0.0398)
10 Years of NRMN	0.1545	0.0785	-0.0010	0.3158	0.1344	-0.3411	0.0492	0.0603	0.0185
	(0.1032)	(0.0677)	(0.0601)	(0.3388)	(0.2402)	(0.3839)	(0.4685)	(0.0828)	(0.0655)
Z	1,230	1,230	1,230	423	251	188	807	679	1,042
Note: Standard errors are	in parentheses								
* 0.10, ** 0.05, and *** 0.0	1 denote signific	cance levels.							

Parameters	Allin	nmigrant work	ers		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.0423	0.0579	0.0487	0.1302	0.1119	0.2343	0.0340	0.0601	0.0577
	(0.0613)	(0.0647)	(0.0631)	(0.1642)	(0.1854)	(0.2139)	(0.0713)	(0.0732)	(0.0701)
5 Years of GE	0.0613	0.0796	0.0696	0.1705	0.1479	0.3446	0.0530	0.0826	0.0818
	(0.0834)	(0.0881)	(0.0860)	(0.2212)	(0.2460)	(0.2852)	(0.0973)	(0.0993)	(0.0953)
10 Years of GE	0.0824	0.0906	0.0829	0.1500	0.1623	0.4883	0.0874	0.0911	0.0930
	(0.1023)	(0.1070)	(0.1046)	(0.2588)	(0.2803)	(0.3233)	(0.1204)	(0.1192)	(0.1148)
3 Years of OE	0.2438***	0.2027***	0.2390***	0.1312	0.2045	0.2030	0.4011*	0.0811	0.1384
	(0.0406)	(0.0499)	(0.0517)	(0.1702)	(0.2173)	(0.2467)	(0.2423)	(0.1263)	(0.1223)
5 Years of OE	0.3249***	0.2753***	0.3170***	0.1838	0.2857	0.2550	0.5252	0.0791	0.1467
	(0.0511)	(0.0579)	(0.0601)	(0.2193)	(0.2640)	(0.2964)	(0.7759)	(0.1918)	(0.1956)
10 Years of OE	0.3598***	0.3231***	0.3559***	0.2491	0.3471	0.2623	-1.8924	0.0520	0.1421
	(0.0657)	(0.0691)	(0.0792)	(0.2275)	(0.2343)	(0.2472)	(6.0921)	(0.2221)	(0.3705)
3 Years of NRCA	0.0974***	0.1193***	0.0819***	0.0540	0.0802	-0.0774	0.3670*	0.4823***	0.1625**
	(0.0349)	(0.0398)	(0.0265)	(0.0458)	(0.0545)	(0.0553)	(0.1877)	(0.1485)	(0.0735)
5 Years of NRCA	0.1509***	0.1891***	0.1320***	0.0735	0.0963	-0.1754*	0.5738	0.6747***	0.2585**
	(0.0551)	(0.0623)	(0.0425)	(0.0895)	(0.0913)	(0.1007)	(0.3603)	(0.2191)	(0.1140)
10 Years of NRCA	0.2305**	0.3050***	0.2177***	0.0633	-0.0232	-0.5464**	0.8697	0.4930	0.4028**
	(0.0915)	(0.0957)	(0.0753)	(0.2778)	(0.2112)	(0.2492)	(4.3292)	(0.3183)	(0.1679)
3 Years of NRCI	0.0162	-0.0716	-0.1577**	0.0280	-0.1441	-0.3815**	-0.0802	-0.0096	-0.0705
	(0.0333)	(0.0449)	(0.0738)	(0.0463)	(0.0875)	(0.1541)	(0.2275)	(0.0520)	(0.1040)
									(Continued)

**Table S5** OLS estimates of cumulative returns to source country GE, OE, and TE (quartic): high-skilled workers

Table S5 Continued									
Parameters	Allin	ımigrant worke	irs		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 Years of NRCI	0.0281	-0.1128	-0.2451**	0.0399	-0.2501*	-0.5796**	-0.7691	-0.0165	-0.1131
	(0.0528)	(0.0704)	(0.1127)	(0.0825)	(0.1399)	(0.2328)	(0.4741)	(0.0809)	(0.1602)
10 Years of NRCI	0.0630	-0.1668	-0.3592**	0.0427	-0.5063**	-0.7201**	10.9744	-0.0271	-0.1993
	(0.0867)	(0.1076)	(0.1536)	(0.2151)	(0.2573)	(0.3350)	(8.7966)	(0.1203)	(0.2293)
3 Years of RCOG	-0.1492***	0.0115	-0.0114	$-0.1664^{***}$	-0.0353	-0.0696	-0.2244	0.0343	-0.0110
	(0.0433)	(0.0413)	(0.0493)	(0.0579)	(0.0772)	(0.3235)	(0.1660)	(0.0619)	(0.0568)
5 Years of RCOG	-0.2445***	0.0179	-0.0197	-0.2632***	-0.0480	-0.0967	-0.4034	0.0589	-0.0179
	(0.0678)	(0.0623)	(0.0783)	(0.0892)	(0.1191)	(0.4602)	(0.4411)	(0.0941)	(0.0903)
10 Years of RCOG	-0.4438***	0.0316	-0.0472	-0.4201***	-0.0254	-0.0584	0.6228	0.1290	-0.0359
	(0.1042)	(0.0851)	(0.1260)	(0.1373)	(0.2005)	(0.4174)	(4.3398)	(0.1245)	(0.1456)
3 Years of RMAN	-0.4502***	$-0.3661^{***}$	-0.4460***	-0.3804**	-0.3015	-1.1005	-0.5958***	-0.3799**	-0.4468***
	(0.1383)	(0.0750)	(0.1238)	(0.1640)	(0.1998)	(0.8093)	(0.2085)	(0.1478)	(0.1422)
5 Years of RMAN	-0.6460***	-0.5367***	-0.6527***	-0.5204**	-0.4058	-1.5851	$-0.9189^{*}$	-0.5363**	-0.5378***
	(0.1986)	(0.1131)	(0.1772)	(0.2375)	(0.2865)	(1.1642)	(0.4862)	(0.2090)	(0.1701)
10 Years of RMAN	-0.5692***	-0.5387***	-0.6689***	-0.2808	-0.1717	-1.4468	0.7358	-0.3697	-0.2453
	(0.2193)	(0.1488)	(0.1997)	(0.2893)	(0.3203)	(1.0872)	(4.3437)	(0.2373)	(0.1909)
3 Years of NRMN	-0.4705***	$-0.1811^{***}$	-0.2608***	$-0.5581^{***}$	-0.3921***	-0.3708	-0.4584***	-0.1374**	-0.2441***
	(0.0912)	(0.0570)	(0.0632)	(0.1313)	(0.1247)	(0.2681)	(0.1676)	(0.0695)	(0.0758)
5 Years of NRMN	-0.7159***	-0.2863***	-0.4073***	-0.8678***	-0.6327***	-0.5690	-0.5130	$-0.2160^{**}$	-0.3776***
	(0.1344)	(0.0894)	(0.0974)	(0.2000)	(0.1989)	(0.4329)	(0.5157)	(0.1087)	(0.1163)
10 Years of NRMN	-0.9025***	-0.4533***	-0.6119***	-1.2103***	-1.0613***	-0.8278	1.3055	-0.3355**	-0.5412***
	(0.1434)	(0.1346)	(0.1379)	(0.2992)	(0.3252)	(0.7594)	(5.0752)	(0.1630)	(0.1624)
N	1,229	1,229	1,229	583	443	363	646	786	866
<i>Note</i> : Standard errors * 0.10, ** 0.05, and ***	are in parenthes 0.01 denote signi	es. ficance levels.							

Parameters	Allim	imigrant worke	ŝrs		Stayers			Switchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.0180	0.0266	0.0307	0.0277	-0.0723	0.2615	0.0658	0.0145	0.0204
	(0.0411)	(0.0417)	(0.0414)	(0.1459)	(0.1544)	(0.1816)	(0.0462)	(0.0444)	(0.0434)
5 Years of GE	0.0310	0.0431	0.0489	0.0393	-0.0863	0.3481	0.0993	0.0284	0.0368
	(0.0558)	(0.0572)	(0.0566)	(0.1943)	(0.2050)	(0.2401)	(0.0634)	(0.0610)	(0.0595)
10 Years of GE	0.0593	0.0732	0.0784	0.0345	-0.0617	0.3828	0.1545**	0.0627	0.0704
	(0.0669)	(0.0702)	(0.0689)	(0.2188)	(0.2326)	(0.2718)	(0.0780)	(0.0753)	(0.0726)
3 Years of OE	0.1531***	0.1718***	0.2031***	0.0287	0.3280**	-0.1074	0.6426***	0.0545	0.0597
	(0.0305)	(0.0380)	(0.0444)	(0.1528)	(0.1474)	(0.1829)	(0.2021)	(0.0819)	(0.1031)
5 Years of OE	0.2035***	0.2281***	0.2722***	0.0424	0.4316**	-0.1012	$1.6618^{**}$	-0.0088	0.0935
	(0.0391)	(0.0469)	(0.0536)	(0.1986)	(0.1884)	(0.2280)	(0.6835)	(0.1130)	(0.1837)
10 Years of OE	0.2266***	0.2423***	0.2989***	0.0715	0.4554**	0.0150	9.1869**	0.0798	0.1186
	(0.0494)	(0.0512)	(0.0601)	(0.2082)	(0.1905)	(0.2178)	(4.5871)	(0.1503)	(0.2858)
3 Years of NRCA	0.1756***	0.1786***	0.1396***	0.1183**	0.0969*	0.0679	0.5889***	0.4800***	0.3998***
	(0.0386)	(0.0418)	(0.0320)	(0.0510)	(0.0561)	(0.0565)	(0.2100)	(0.1504)	(0.1181)
5 Years of NRCA	0.2747***	0.2827***	0.2214***	0.1815**	0.1395	0.1016	0.1511	0.7524***	0.5892***
	(0.0603)	(0.0650)	(0.0489)	(0.0855)	(0060.0)	(0.0941)	(0.2535)	(0.1960)	(0.1632)
10 Years of NRCA	0.4220***	0.4513***	0.3505***	0.2626	0.1489	0.1317	-6.7602*	0.9597**	0.6208***
	(0.0937)	(0.0972)	(0.0718)	(0.1768)	(0.1697)	(0.2001)	(3.4645)	(0.4552)	(0.1697)
3 Years of NRCI	0.0880**	-0.0595	-0.0809*	0.1123**	-0.2228***	-0.2877**	0.0174	0.0830*	-0.0588
	(0.0412)	(0.0431)	(0.0426)	(0.0478)	(0.0753)	(0.1180)	(0.2173)	(0.0492)	(0.0627)
									(Continued)

Table S6 Continued									
Parameters	All im	migrant worke	ers		Stayers		0	Switchers	
I	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 Years of NRCI	0.1356**	-0.0955	-0.1306*	0.1597**	-0.3620***	-0.4878**	-0.7791*	0.1176	-0.0953
	(0.0635)	(0.0655)	(0.0677)	(0.0757)	(0.1218)	(0.2057)	(0.4380)	(0.0755)	(0.1000)
10 Years of NRCI	0.1865**	-0.1547*	-0.2281**	0.1387	-0.6065***	-0.9448**	-7.7002**	0.0908	-0.1704
	(0.0893)	(0.0883)	(0.1100)	(0.1291)	(0.2201)	(0.4450)	(3.6392)	(0.1038)	(0.1644)
3 Years of RCOG	-0.0691**	0.0718**	-0.0321	-0.0939***	0.0547	-0.0772	-0.3377***	0.1110*	-0.0010
	(0.0307)	(0.0349)	(0.0345)	(0.0344)	(0.0573)	(0.0958)	(0.1301)	(0.0634)	(0.0427)
5 Years of RCOG	$-0.1161^{**}$	0.1115**	-0.0519	$-0.1490^{**}$	0.0995	-0.1288	-1.0500***	0.1792*	-0.0016
	(0.0485)	(0.0550)	(0.0549)	(0.0606)	(0.0933)	(0.1539)	(0.3978)	(0.0965)	(0.0679)
10 Years of RCOG	-0.2268***	0.1614*	-0.0877	-0.2404	0.2271	-0.2512	-6.7981**	0.3069**	-0.0022
	(0.0767)	(0.0860)	(0.0800)	(0.1619)	(0.1864)	(0.2990)	(3.3230)	(0.1304)	(0.1092)
3 Years of RMAN	-0.0969**	$-0.1025^{**}$	$-0.1291^{**}$	-0.0441	-0.1279	0.4435	$-0.4116^{***}$	-0.0773	-0.1443**
	(0.0437)	(0.0451)	(0.0580)	(0.0612)	(0.0928)	(0.5220)	(0.1274)	(0.0578)	(0.0592)
5 Years of RMAN	-0.1507**	$-0.1560^{**}$	$-0.1954^{**}$	-0.0480	-0.1715	0.7323	-1.1908***	-0.1221	-0.2212**
	(0.0693)	(0.0706)	(0.0902)	(0.1013)	(0.1480)	(0.8107)	(0.4040)	(0.0881)	(0.0896)
10 Years of RMAN	$-0.2158^{*}$	-0.2072*	-0.2399*	0.0641	-0.0787	1.3198	-7.1982**	-0.2003	-0.2970**
	(0.1114)	(0.1072)	(0.1362)	(0.2030)	(0.2539)	(1.1685)	(3.4165)	(0.1232)	(0.1327)
3 Years of NRMN	-0.0475	-0.0337	-0.0595**	-0.0090	-0.0168	0.0084	-0.3276***	-0.0213	-0.0580*
	(0.0369)	(0.0276)	(0.0302)	(0.0543)	(0.0795)	(0.1077)	(0.1232)	(0.0328)	(0.0339)
5 Years of NRMN	-0.0702	-0.0563	-0.0964**	-0.0083	-0.0396	0.0112	-1.0685***	-0.0358	-0.0920*
	(0.0597)	(0.0442)	(0.0475)	(0.0956)	(0.1252)	(0.1712)	(0.4053)	(0.0527)	(0.0527)
10 Years of NRMN	-0.0867	-0.1128	-0.1717**	0.0252	-0.1458	-0.0054	-7.0638**	-0.0728	-0.1499*
	(0.1044)	(0.0738)	(0.0741)	(0.2303)	(0.2056)	(0.3009)	(3.4280)	(0680.0)	(0.0805)
N	1,875	1,875	1,875	779	532	426	1,096	1,343	1,449
<i>Note</i> : Standard errors ar <sup>1</sup> * 0.10, ** 0.05, and *** 0.0	e in parenthese )1 denote signif	s. icance levels.							

Parameters	All in	nmigrant worke	srs		Stayers		S	witchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
3 Years of GE	0.1760*	0.1846*	0.1500	0.4061*	0.3633	0.1398	0.0831	0.1185	0.0798
	(0.0923)	(0.1041)	(0.0948)	(0.2311)	(0.4090)	(0.6474)	(0.1229)	(0.1230)	(0.1103)
5 Years of GE	0.2305*	0.2469*	0.2016	$0.5301^{*}$	0.5082	0.2693	0.1101	0.1547	0.1022
	(0.1254)	(0.1400)	(0.1287)	(0.3123)	(0.5486)	(0.8239)	(0.1676)	(0.1646)	(0.1489)
10 Years of GE	0.2472	0.2745*	0.2245	0.5553	0.6785	0.6249	0.1226	0.1559	0.0898
	(0.1515)	(0.1626)	(0.1540)	(0.3731)	(0.6325)	(0.8101)	(0.2040)	(0.1881)	(0.1751)
3 Years of OE	0.0987*	0.1500**	0.0937	0.0756	0.5110	0.5710	0.6353*	0.1689	-0.0209
	(0.0526)	(0.0655)	(0.0654)	(0.2357)	(0.3857)	(0.4987)	(0.3402)	(0.1401)	(0.2099)
5 Years of OE	0.1441**	0.2115***	0.1642**	0.1103	0.6086	0.6925	1.1647**	0.1131	-0.0394
	(0.0673)	(0.0754)	(0.0764)	(0.3068)	(0.4993)	(0.6220)	(0.5758)	(0.2205)	(0.2622)
10 Years of OE	0.1960**	0.2680***	0.3069***	0.1567	0.4459	0.5522	-7.5834	0.1198	-0.0221
	(0.0812)	(0.0816)	(0.0874)	(0.3279)	(0.5255)	(0.6007)	(6.6156)	(0.2012)	(0.2296)
3 Years of NRCA	0.1723***	0.1410**	0.1517***	0.2172**	0.0608	0.0792	-0.0269	0.3468	0.1511
	(0.0523)	(0.0672)	(0.0533)	(0.0888)	(0.1396)	(0.1496)	(0.3994)	(0.2272)	(0.1445)
5 Years of NRCA	0.2676***	0.2311**	0.2454***	0.3888**	0.1367	0.1243	0.1969	0.4681	0.2497
	(0.0836)	(0.1043)	(0.0851)	(0.1783)	(0.2369)	(0.2941)	(1.0138)	(0.3039)	(0.2300)
10 Years of NRCA	0.4144***	0.4183***	0.4202***	0.8856	0.4486	0.1879	0.7902	0.2194	0.4688
	(0.1432)	(0.1583)	(0.1434)	(0.5842)	(0.5449)	(0.8443)	(3.9783)	(0.3063)	(0.3723)
3 Years of NRCI	0.1345**	0.0440	-0.0247	0.1439	0.0879	-0.1406	-0.3496	-0.0334	-0.1156
	(0.0582)	(0.0606)	(0.0850)	(0.0889)	(0.1111)	(0.1372)	(0.3881)	(0.0898)	(0.1134)
									(Continued)

 Table S7
 OLS estimates of cumulative returns to source country GE, OE, and TE (quartic): low income source countries

Table S7 Continued									
Parameters	Allin	nmigrant work	ers		Stayers		S	witchers	
	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit	1-Digit	2-Digit	3-Digit
5 Years of NRCI	0.2115**	0.0794	-0.0282	0.2344	0.1776	-0.2412	-2.4397	-0.0488	-0.1663
	(0.0926)	(0.0951)	(0.1330)	(0.1420)	(0.1814)	(0.2131)	(2.4121)	(0.1405)	(0.1778)
10 Years of NRCI	0.3247**	0.1974	0.0336	0.3979	0.5180	-0.4738	-71.7340	-0.0476	-0.1348
	(0.1511)	(0.1490)	(0.2000)	(0.2515)	(0.4038)	(0.3974)	(121.2068)	(0.2160)	(0.2685)
3 Years of RCOG	0.0276	0.0365	0.1361*	0.0657	0.0443	0.5681*	-0.3210	-0.0994	0.0686
	(0.0499)	(0.0562)	(0.0766)	(0.0715)	(0.1214)	(0.3288)	(0.2633)	(0.1271)	(0.0818)
5 Years of RCOG	0.0412	0.0519	0.2159*	0.0976	0.0502	0.8328*	-0.9807*	-0.1457	0.1130
	(0.0789)	(0.0889)	(0.1178)	(0.1130)	(0.1990)	(0.4768)	(0.5243)	(0.1921)	(0.1253)
10 Years of RCOG	0.0554	0.0487	0.3447**	0.1191	-0.0423	0.8509	3.5206	-0.1366	0.2125
	(0.1251)	(0.1412)	(0.1620)	(0.1890)	(0.3736)	(0.5677)	(3.6925)	(0.2496)	(0.1678)
3 Years of RMAN	-0.0849	-0.2065***	-0.1132	-0.2363	-0.6024*	-1.2319***	-0.2865	-0.2539	-0.0172
	(0.0705)	(0.0704)	(0.0823)	(0.2273)	(0.3363)	(0.4279)	(0.2435)	(0.1583)	(0.1004)
5 Years of RMAN	-0.1167	$-0.3115^{***}$	-0.1732	-0.3285	-0.7913	-1.8159***	$-1.0651^{**}$	-0.3592*	-0.0099
	(0.1077)	(0.1061)	(0.1242)	(0.3321)	(0.4966)	(0.6387)	(0.5105)	(0.2123)	(0.1522)
10 Years of RMAN	-0.0699	$-0.3810^{***}$	-0.2265	-0.2156	-0.2468	-1.9637**	3.3255	-0.2931*	0.1144
	(0.1500)	(0.1407)	(0.1630)	(0.4088)	(0.6613)	(0.9023)	(3.6683)	(0.1712)	(0.2037)
3 Years of NRMN	-0.2291**	0.0164	-0.0255	-0.5392***	-0.0428	-0.1844	-0.4149	0.0315	0.0074
	(0.1133)	(0.0358)	(0.0378)	(0.1923)	(0.1098)	(0.1508)	(0.2724)	(0.0473)	(0.0452)
5 Years of NRMN	-0.3224**	0.0298	-0.0381	-0.8002***	-0.1575	-0.2577	-0.6793	0.0542	0.0130
	(0.1537)	(0.0577)	(0.0607)	(0.2754)	(0.1932)	(0.2623)	(0.4553)	(0.0742)	(0.0715)
10 Years of NRMN	-0.2811	0.0660	-0.0518	-0.9859	-0.7656	-0.2248	-1.2970	0.0997	0.0210
	(0.3164)	(0.0992)	(0.1034)	(0.6329)	(0.4689)	(0.6095)	(0.9176)	(0.1156)	(0.1151)
Z	584	584	584	227	162	125	357	422	459
<i>Note</i> : Standard errors * 0.10, ** 0.05, and ***	are in parenthese 0.01 denote signif	s. icance levels.							